

Contents lists available at ScienceDirect

### Journal of Cleaner Production



journal homepage: www.elsevier.com/locate/jclepro

# Motivations for voluntary corporate adoption of integrated reporting: A novel context for comparing voluntary disclosure and legitimacy theory

Kimitaka Nishitani<sup>a,\*</sup>, Jeffrey Unerman<sup>b</sup>, Katsuhiko Kokubu<sup>c</sup>

<sup>a</sup> Research Institute for Economics and Business Administration, Kobe University, 2-1 Rokkodai Nada Kobe, 657-8501, Japan

<sup>b</sup> Department of Accounting and Finance, Lancaster University Management School, Lancaster, LA1 4YX, United Kingdom

<sup>c</sup> Graduate School of Business Administration, Kobe University, 2-1 Rokkodai Nada Kobe, 657-8501, Japan

#### ARTICLE INFO

Handling Editor: Dr. Govindan Kannan

Keywords: Integrated reporting Sustainability reporting Voluntary reporting motivations Voluntary disclosure theory Legitimacy theory

#### ABSTRACT

The swift development of integrated reporting as a genre within corporate reporting and the rapid spread of organizations adopting the International Integrated Reporting Council's (IIRC) vision for such shows that corporate voluntary reporting is now moving to a new stage. This provides an opportunity for companies to refine or reveal their hitherto hidden voluntary reporting philosophy by adapting their reporting practices. The motivations for adopting integrated reporting could differ from the legitimation motivations the literature has previously considered to underlie broader social and environmental sustainability reporting. For this reason, this study explores the motivations for companies to shift to integrated reporting or to continue with social and environmental sustainability reporting in terms of voluntary disclosure and legitimacy theory. The analysis employs survival analysis conducted using data from companies listed on the Tokyo and London stock exchanges to explore the relationships between social and environmental performance and the timing and duration of a shift to integrated reporting. The results reveal different motivations underlying voluntary reporting practices between the two countries and across industry sectors. Overall, the findings suggest that the financial transparency and accountability role of integrated reporting proposed by the IIRC has gained traction among Tokyo but not London listed companies. However, there is a trend towards accountability in the shift to integrated reporting consistent with voluntary disclosure theory even for London listed companies at the industry sector level. This supports the view that companies have an incentive to differentiate their voluntary reporting philosophy through different forms of reporting and that legitimacy theory and voluntary disclosure theory are compatible in this regard.

#### 1. Introduction

A rich academic literature examining the motivations for voluntary corporate reporting has developed in recent decades (Bebbington et al., 2014; Deegan and Unerman, 2011), mostly concerned with evolving social and environmental sustainability reporting practices and legitimacy theory (Alrazi et al., 2015; Lindblom, 1993; Suchman, 1995). The background to this literature is that industrialization has negatively impacted society and the natural environment. This results in the increasing demand for sustainability (Awan et al., 2018), defined as "[meeting] the needs of the present without compromising the ability of future generations to meet their own needs" (Brundtland, 1987, para 27), with an emerging consensus that sustainability incorporates three

dimensions, namely, the social, the environmental, and the economic (Haugh and Talwar, 2010). Because companies have a corporate social responsibility (CSR) combined with specific skills and resources for innovation in society and the environment, the expectation is that they will play a primary role in achieving sustainability (Cheng et al., 2021; Scheyvens et al., 2016). In response, many companies link sustainability to their business.<sup>1</sup> Within such an explanatory framework, the motivation for voluntary social and environmental sustainability disclosures (e. g., CSR reporting) represent a managerial desire to build, support, or repair the legitimacy with which an array of politically and economically powerful stakeholders regard a reporting organization (Deegan, 2002, 2014).

Long dismissed by these studies is the alternative of voluntary

\* Corresponding author.

https://doi.org/10.1016/j.jclepro.2021.129027

Received 28 March 2021; Received in revised form 23 August 2021; Accepted 13 September 2021 Available online 20 September 2021

0959-6526/© 2021 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

E-mail address: kimitakan@rieb.kobe-u.ac.jp (K. Nishitani).

<sup>&</sup>lt;sup>1</sup> For example, a varied degree of competencies in the product life cycle is required to ensure and improve environmental performance (Awan et al., 2020).

disclosures motivated by a managerial desire to be more fully financially accountable to investors, namely, shareholders and long-term creditors, although no social, environmental, or economic sustainability, even at the company level, can be intrinsically sacrificed to achieve full sustainability (Awan et al., 2019; Elkington, 2004). Thus, the role of corporate voluntary reporting is if anything to secure legitimacy, and so the practices and motivations to discharge financial accountability to investors differ from the social, environmental, and economic accountability to a broader range of stakeholders (Adams, 2004; Arunachalam and McLachlan, 2015).

Over the past decade, following an expansion in environmental, social, and governance investment and the foundation in 2010 of the International Integrated Reporting Committee, later renamed the International Integrated Reporting Council (IIRC), a new voluntary reporting initiative for integrated reporting has rapidly gained traction (de Villiers et al., 2014). The IIRC's vision of integrated reporting seeks to incorporate the reporting of the social, environmental, and economic sustainability factors that are of most relevance to an organization's long-term financial performance, primarily to help inform the investment decisions made by long-term investors (Humphrey et al., 2017).

Thus, integrated reporting can improve the quality of information in terms of covering all three dimensions of sustainability at the company level. However, while integrated reporting overlaps with important broader aspects of social and environmental sustainability reporting, there is a key difference in that there is a more explicit and narrow focus on long-term financial stakeholders (investors), not on merely securing legitimacy from other stakeholders. In doing so, integrated reporting supplies information about social and environmental impacts likely to be material to an organization's longer-term financial results, rather than the broader range of impacts addressed in social and environmental sustainability reporting.

The rapid development of integrated reporting as a new genre within corporate reporting, and the rapid spread of organizations adopting the IIRC's vision of integrated reporting suggests that corporate voluntary reporting is now moving to a new stage (Correa-Garcia et al., 2020; Perego et al., 2016). This may provide an opportunity for companies, especially those with differing motivations for voluntary reporting, to refine or reveal their hitherto (hidden) voluntary reporting philosophy by adapting their reporting practices. The coexistence of the IIRC's framework for integrated reporting and Global Reporting Initiative (GRI) standards for social and environmental sustainability reporting supports such a refinement in the demand for both integrated reporting and social and environmental sustainability reporting.

Because companies with motivations to discharge financial accountability to investors could not previously reveal their reporting philosophy through social and environmental sustainability reporting, the potential motivations (other than rhetorical legitimation) underlying voluntary reporting may now become clear. Indeed, data collected as part of this study suggest that many companies replace social and environmental sustainability reporting with integrated reporting, rather than adopting integrated reporting in addition to social and environmental sustainability reporting. This may be a sign of the differentiation of reporting philosophy. If not, there is a potential distortion of the primary purpose of integrated reporting by its users, suggesting that the IIRC's vision for integrated reporting is as yet unresolved.

Unfortunately, there is currently little evidence addressing this, with existing studies analysing social and environmental sustainability reporting from a legitimacy perspective and only clarifying part of the role of voluntary corporate reporting. Because these account for most studies analysing voluntary corporate reporting in accounting research and given the equal lack of attention to the financial accountability role, they may have misinterpreted the role of voluntary corporate reporting. Indeed, the legitimating role clarified by earlier studies cannot address the reality that environmental, social, and governance information has recently become as important as financial information for investment decisions, especially in the era of the United Nations' Sustainable Development Goals (Nishitani et al., 2020). This implies an underestimation of the required role of company voluntary reporting by the literature, in the sense that the financial accountability and legitimating roles should not intrinsically conflict.

To address this gap in the research, this study examines voluntary reporting philosophy and practices in a unique environment, and in doing so supplies novel insights about the hidden motivations underlying voluntary reporting. In developing these insights, the goal is to learn the apparent primary motivation for companies to shift to integrated reporting or to continue with social and environmental sustainability reporting. This enables us to expose the hidden voluntary reporting philosophy of companies, which they reveal by adapting their reporting practices. This serves as the primary novelty of this study.

In our analysis, we consider that a company's voluntary reporting philosophy is expressed in a combination of: (i) a managerial desire to be financially accountable to the primary target audience of integrated reports, namely, investors, and (ii) managerial concerns over ensuring legitimacy with a broader range of politically and economically powerful stakeholders. To help learn the motivations derived for companies to shift to integrated reporting or continue with social and environmental sustainability reporting, we assess the consistency of these hypotheses with both voluntary disclosure theory and legitimacy theory following existing studies in the area. For example, de Villiers and van Staden (2011) found that companies used different reporting media (i.e., annual reports or websites) depending on their environmental performance.

The hypothesis relating to voluntary disclosure theory could reveal financial accountability to investors as the motivation for engaging in voluntary reporting (e.g., Al-Tuwaijri et al., 2004; Bewley and Li, 2000; Clarkson et al., 2008; Cormier and Magnan, 1999; Dye, 1985; Lang and Lundholm, 1993; Magness, 2006; Verrecchia, 1983). In contrast, the hypothesis consistent with legitimacy theory could suggest that rhetorical legitimation to a broader range of politically and economically powerful stakeholders is the primary motivation for voluntary reporting (e.g., Cho and Patten, 2007; Cho et al., 2012; Clarkson et al., 2011; Deegan, 2002; Gray et al., 1995; Guidry and Patten, 2012; O'Donovan, 2002; Patten, 2002; Wilmshurst and Frost, 2000).

In turn, where companies have proactively incorporated social and environmental factors into their strategy for improving long-term financial performance and understand and manage the interrelating risks and opportunities, the motivation to shift to integrated reporting would be consistent with voluntary disclosure theory (Stacchezzini et al., 2016). At the same time, where companies continue with existing social and environmental sustainability reporting, the motivation for voluntary reporting would be consistent with legitimacy theory (as supported by the extant literature). Otherwise, integrated reporting could be just a developed form of existing social and environmental sustainability reporting (and thereby deviating from its primary purpose). From a legitimacy perspective, "[integrated reporting] is a [discretionary] disclosure choice and, as such, it may represent a reporting [initiative] to manage corporate legitimacy" (Lai et al., 2016, p.165).

According to the literature, voluntary disclosure theory predicts a positive relationship between social and environmental performance and its reporting, while legitimacy theory envisages a negative relationship. This suggests that social and environmental performance is an important determinant of corporate voluntary reporting in both theories. Thus, using both theories to analyse the relationships between social and environmental performance and integrated or social and environmental sustainability reporting can help clarify the roots of the movement in corporate voluntary reporting in a novel context.

Although earlier empirical studies suggest that legitimacy theory is a more reasonable explanation of the role of existing social and environmental sustainability reporting practices, companies can refine or uncover their (hidden) voluntary reporting philosophy between financial accountability and rhetorical legitimation by adapting their reporting practices. Indeed, voluntary disclosure theory would be a more reasonable explanation of the role of integrated reporting where the motivation of a company for financial accountability to its investors is a commitment to the principles underlying the IIRC's integrated reporting framework. As such, the focus of this study is more on the reporting philosophy than on the mere adoption of integrated or social and environmental sustainability reporting. Fig. 1 portrays the conceptual framework applied in the analysis.

To assess the motivations underlying any shift to integrated reporting compared with a continuation of social and environmental sustainability reporting, we conduct a survival analysis of the relationship between social and environmental performance and the hazard rate. The hazard rate here is the probability of the timing and duration of a company shifting to integrated reporting where social and environmental sustainability reporting was the company's usual reporting practice. Thus, the survival analysis is not just about whether but when a company adopts integrated reporting. As companies decide when to adopt integrated reporting, they have an incentive to continue with it, regardless of whether they replace or compensate sustainability reporting with integrated reporting. This is consistent with the argument that it is unrealistic for companies to frequently change their voluntary reporting philosophy. The intrinsic determinant is the timing of first adoption (Nishitani, 2009) not adoption per se, and therefore the limited dependent variable models (probit and logit) common in the literature are not suitable.

For our analysis, we collect data from the 100-largest companies listed on each of the Tokyo and London stock exchanges from 2012 to 2017 and correct for any latent selection bias associated with sampling only the largest companies in each market using the inverse Mills ratio as the correction term. The motivation for the sample selection of Tokyo and London listed companies is the longstanding tradition of corporate social and environmental sustainability reporting (and the underlying engagement with CSR) in both countries. As we expect integrated reporting to reflect, at least in part, the consequences of a company's economic performance on its social and environmental impact, its early development and adoption in Japan and the UK could draw upon an existing sustainability reporting skills base that have made it quick and easy to implement.

In contrast, in, say, the US where there is a much less developed tradition of social and environmental sustainability reporting (Dilling and Harris, 2018), we expect integrated reporting to require much greater advances in the underlying reporting skills, and thus a potentially longer gestation period. Further, while there are major corporate cultural differences between the two countries, such as those in corporate governance systems, we consider that a comparison between Japanese and UK integrated reporting practices is proper given the similarities in context as developed market-based economies (Hofstede, 2001, but also Frias-Aceituno et al., 2013; McSweeney, 2002a, 2002b; Yonekura et al., 2012). Lastly, Japanese reporting practices have the unique feature that social and environmental sustainability reporting in

annual reports has never been common, and this serves as a suitable contrast with experience in the UK.

The findings show that there are different motivations underlying voluntary reporting practices both between the two countries and across industry sectors. Overall, it would appear that the primary sustainability-related financial transparency and accountability role of integrated reporting proposed by the IIRC has gained traction among Tokyo but not London listed companies. However, we also reveal a trend toward an accountability motivation in the shift to integrated reporting consistent with voluntary disclosure theory, even for London listed companies at the industry sector level. These findings support the view that companies have an incentive to differentiate their voluntary reporting philosophy by different forms of reporting, and that legitimacy theory and voluntary disclosure theory are compatible in this regard.

The remainder of the paper is organized as follows. Section 2 provides an overview of integrated reporting. Section 3 presents the theoretical framework, reviews the literature, and develops the hypotheses. Section 4 explains the research design and Section 5 details the empirical results. Section 6 presents the conclusion, discusses some limitations of the study, and suggests some directions for future research.

#### 2. Development of integrated reporting

Since the beginning of the 21st century, *sustainability reporting* has become a term widely used to describe the self-reporting of a company's social and environmental policies, practices, and performance, including any associated risks and opportunities. Although we can trace elements of social and environmental reporting back as far as the 19th century (Guthrie and Parker, 1989), social and environmental reporting became much more common towards the end of the 20th century. Developments in social and environmental reporting in the 1990s saw an increasing number of companies devoting part of their annual financial report to the disclosure of information on social and environmental matters affecting the company. As social and environmental reporting became more sophisticated, and companies addressed a wider range of issues within their reporting, the space devoted to social and environmental information within the annual financial report expanded.

Increasingly, companies began to publish social and environmental information in separate reports, with these reports now commonly known as sustainability reports (de Villiers et al., 2014). This enabled the annual financial report to focus primarily on financial performance and risk information aimed at investors, with minimal social and environmental sustainability information. In contrast, the sustainability report became longer and more detailed, and tended to aim at addressing the information needs of a broad range of politically and economically powerful stakeholders (Jensen and Berg, 2012). While sustainability reports have now become a common medium for sustainability reporting, some companies have continued to use annual reports to reveal their social and environmental impact information (Michelon et al., 2015).



Fig. 1. Motivations for the shift to integrated reporting.

Over the past few years, advocates of integrated reporting have argued that the decisions of investors would be improved by the provision within a single primary corporate report of concise information integrating a company's social, environmental, and financial performance and their impact (Eccles and Krzus, 2010). A key purpose of the development of integrated reporting has been to provide this concise integrated picture (International Integrated Reporting Council IIRC, 2013).

In practice there are three main foundations of integrated reporting. At the organizational level, the Danish pharmaceutical company Novo Nordisk pioneered an integrated approach with its own reporting in the early years of the 21st century (de Villiers et al., 2014). At the country level, South Africa pioneered and championed integrated reporting, requiring all companies with shares listed on the Johannesburg Stock Exchange to either publish an integrated report or explain why they had not (Cheng et al., 2014). South African regulations for integrated reporting initially saw it aimed at meeting the information needs of an extremely broad range of stakeholders, not just investors or the most important stakeholders. At the international level, since 2010 the IIRC has championed the development and fostered the rapid and geographically widespread adoption of an integrated reporting framework aimed at meeting the information needs of investors (de Villiers et al., 2014; International Integrated Reporting Committee IIRC, 2011; 2013; Rinaldi et al., 2018). Although only published in 2013, the IIRC first introduced the basic elements of its integrated reporting framework in a 2011 discussion paper (International Integrated Reporting Committee IIRC, 2011; 2013).

Through the IIRC's initiatives, integrated reporting has now become prominent as a new form of (mainly) voluntary corporate reporting for the disclosure of financial and nonfinancial information to investors, including the social and environmental sustainability information likely to affect longer-term financial performance (International Integrated Reporting Council IIRC, 2016). As the IIRC has dominated these developments globally, and given that the experience of Novo Nordisk and South Africa were later and differ from the IIRC's approach to integrated reporting, we consider the IIRC's integrated reporting framework to be most relevant to the aims of this study.

According to the IIRC, the widespread adoption of its vision for integrated reporting is expected to:

[1] improve the quality of information available to providers of financial capital to enable a more efficient and productive allocation of capital; [2] promote a more cohesive and efficient approach to corporate reporting that draws on different reporting strands and communicates the full range of factors that materially affect the ability of an organization to create value over time; [3] enhance accountability and stewardship for the broad base of capitals (financial, manufactured, intellectual, human, social and relationship, and natural) and promote understanding of their interdependencies; [4] support integrated thinking, decision-making and actions that focus on the creation of value over the short, medium and long term (International Integrated Reporting Council IIRC, 2013, p.2).

In this manner, integrated reporting serves as a new form of reporting to address the suggested weakness of narrower financial reporting, such as a lack of responsiveness to new value drivers and a changing business context (Adams and Simnett, 2011; Stubbs and Higgins, 2014). Thus, the IIRC expects integrated reporting to contribute to improving the longer-term economic/financial decision-making of investors.

## 3. Theoretical and empirical literature and hypothesis development

This section first explains our theoretical framework using voluntary

disclosure theory and legitimacy theory. It then reviews studies that have analysed the related fields of social and environmental sustainability reporting using these theories. Lastly, it draws on insights from theory and practice in developing hypotheses about the motivations for companies shifting to integrated reporting where social and environmental sustainability reporting had been their usual practice.

#### 3.1. Theoretical framework

Voluntary disclosure theory and legitimacy theory are among the most widespread theories used in empirical studies of corporate voluntary reporting (Hahn and Kühnen, 2013). In what follows, we explain the key aspects of voluntary disclosure theory and legitimacy theory. Voluntary disclosure theory as it applies in the social and environmental sustainability reporting literature is rooted in the financial disclosure literature, with Verrecchia (1983) and Dye (1985) arguing about the role of proprietary costs and uncertainty, respectively.

Voluntary disclosure theory was originally a special case of game theory (Dye, 2001) and follows the premise that companies will provide favourable information (i.e., good news) and withhold unfavourable information (i.e., bad news) (Dye, 2001; Verrecchia, 1983). To effectively interpret the motivation of a company making or withholding disclosures, investors should then expect the company has incentives to behave in this manner (Dye, 2001). If investors interpret withholding (or nondisclosure) as a sign of the existence of unfavourable information, the company's share price will fall, so companies with favourable information have an incentive to disclose this information to increase market value (Depoers, 2000; Kent and Ung, 2003).

However, because a proprietary cost associated with disclosing information (i.e., a reduction in future cash flows attributable to the disclosure) can occur where competitors gain benefit from information in the disclosure, investors may be uncertain about whether information is withheld to avoid reporting bad news or to avoid incurring proprietary costs despite the information withheld containing good news (Cheng, 2017; Depoers, 2000; Dye, 1985, 2001, 2001; Skinner, 1994; Verrecchia, 1983).<sup>2</sup> Thus, a proprietary cost increases the range of interpretations for nondisclosure in conditions of investor uncertainty about what has occurred within a company, and there is no longer an unambiguous implication that the withheld information is unfavourable (Craswell and Taylor, 1992).

The nature of these proprietary costs decides the threshold level of disclosure where a company only discloses information above the threshold and otherwise withholds it (Lang and Lundholm, 1993; Verrecchia, 1990). This characterizes a partial disclosure equilibrium in which a company follows a strategy of disclosing only successes. Therefore, a higher (lower) proprietary cost is associated with a lower (higher) level of voluntary disclosure in conditions of investor uncertainty about what has occurred within a company, because the disclosure may increase the risk of being vulnerable to competitors (Rezaee and Tuo, 2017).

Voluntary disclosure theory within the financial reporting literature focuses on the provision of financial rather than social and environmental information, so it has had to be adapted for use in analysing social and environmental sustainability reporting where: (i) there is information asymmetry between companies and investors, and (ii) there are potential proprietary costs associated with the disclosure of social and/or environmental information (Guidry and Patten, 2012). Within the social and environmental sustainability reporting literature (e.g., Al-Tuwaijri et al., 2004; Clarkson et al., 2008) we modify the notions within voluntary disclosure theory to enable empirical statistical analysis in the social and environmental sustainability context, which theorizes that better (worse) environmental performance incurs lower

<sup>&</sup>lt;sup>2</sup> Proprietary costs are a plausible reason for why not all companies disclose information.

(higher) proprietary cost under conditions of uncertainty.

Voluntary disclosure theory as applied to social and environmental sustainability contexts suggests that the motivation for companies to use social and environmental sustainability reporting is to aid improved decision-making by their investors. Voluntary disclosure theory assumes companies that have achieved better financially impactful social and environmental performance have an incentive to disclose this information to differentiate themselves from those that have not had to reduce the asymmetric information between the company and its investors (Clarkson et al., 2008; Li et al., 1997).

This aligns with Lang and Lundholm's (1993, p.247) suggestion that companies have motivations for corporate reporting, such as "... overcoming adverse selection [and] reducing transaction costs in the market". Such behaviour is reasonable where investors evaluate corporate social and environmental performance, not directly, but indirectly through disclosed information, and where the achievement of better social and environmental performance potentially provides economic/financial benefits for the reporting organization. These potentially include an increase in customer demand by showing consideration for society and/or the environment, an improvement in production efficiency via improving management control, and/or a reduction of the environmental risks to which companies may expose themselves in the future (Nishitani, 2011).

Voluntary disclosure theory as applied to social and environmental sustainability disclosures therefore proposes that a company's primary focus for corporate sustainability reporting is on the possible or probable financial performance benefits from its improved social and environmental performance, rather than on broader aspects of social and environmental sustainability (Adams, 2004). That is, financial accountability and transparency through corporate reporting are critical elements required for capital markets to function efficiently (Healy and Palepu, 2001). In practice, providers of financial capital can infer the longer-term financial impacts, risks, and opportunities arising from social and environmental performance, such as lower latent environmental liabilities, by evaluating the information disclosed (Bewley and Li, 2000; Clarkson et al., 2008).

Thus, from the perspective of voluntary disclosure theory, companies that have achieved better financially impactful social and environmental performance can potentially increase their market valuation by disclosing social and environmental information, especially objective types of information that are difficult to mimic by companies that have not achieved such good levels of performance (Al-Tuwaijri et al., 2004; Bewley and Li, 2000; Clarkson et al., 2008). Conversely, financial markets may potentially judge companies that have not disclosed financially beneficial social and environmental sustainability information as "average-type" companies that have not gained financial advantages associated with social and environmental performance (Clarkson et al., 2008).

Given the above arguments, voluntary disclosure theory predicts a positive relationship between social and environmental performance and its reporting (Clarkson et al., 2008, 2011). Where there is such a positive relationship, voluntary disclosure theory suggests that managers disclose social and environmental sustainability information given a desire to provide financial transparency (and accountability) to their investors for enhancing their financially impactful social and environmental performance.

In contrast, legitimacy theory assumes that a company has no inherent right to exist, but this right is conferred upon it by society through a social contract where a company performs various socially desired actions in return for approval of its objectives, other rewards, and its ultimate survival (Deegan, 2002; Guthrie and Parker, 1989; Magness, 2006, 2008):

[A]ny social institution [including a company] operates in society via a social contract, expressed or implied, whereby its survival and growth are based on: (1) the delivery of some socially desirable ends to society in general, and (2) the distribution of economic, social, or political benefits to groups from which it derives its power (Shocker and Sethi, 1973, p.97).

In short, legitimacy theory stresses how a company will react to society's expectations (Wilmshurst and Frost, 2000). Where a company is successful in meeting, or perceived as meeting, its social contract, it leads to a congruence between the company and society (Cormier and Gordon, 2001). Thus, legitimacy theory is reactive (Guthrie and Parker, 1989).

Central to the concept of legitimacy are the perceptions held by relevant parts of the public and by society at large (Aerts and Cormier, 2009). Deephouse and Carter (2005, p.331) explain that a fundamental element of legitimacy is "... meeting and adhering to the expectations of a social system's norms, values, rules, and meanings". Legitimacy theory argues that companies use social and environmental disclosure as a (social) legitimating tool to create an impression (that may or may not reflect underlying performance) that the company operates in a manner that meets the social and environmental expectations (delivering the social contractual rights) of stakeholders who have political and economic power over the company (Lindblom, 1993). Legitimacy theory therefore focuses upon a company's exposure to its social and political environment, rather than its stock market context (Cho et al., 2012; Patten, 1991, 2000). In short, legitimacy theory focuses on how corporate management will react to societal perceptions (Wilmshurst and Frost, 2000).

Because legitimacy theory is based on the notion of a social contract, if politically and economically powerful members of society perceive that a company has breached the implicit terms of its social contract, it may threaten its survival (Cho et al., 2012; Deegan, 2002; Deegan et al., 2002). Indeed, Deegan (2002, p.293) suggests that "... where society is not satisfied that [a company] is operating in an acceptable, or legitimate, manner, then society will effectively revoke [the company's] 'contract' to continue its operations". This could be through consumers reducing their demand for the company's products or services, and/or by nongovernment organizations lobbying for legislation that would impact upon the cash flow of the company (Magness, 2006). Therefore, social and environmental disclosure may be used "... to reinforce the [society's] perception of management's responsiveness to specific [social and] environmental issues, or alternatively to divert attention from adverse [social and] environmental situations" (Wilmshurst and Frost, 2000, p.11).

According to Lindblom (1993), from the perspective of environmental legitimacy, companies may use environmental reporting as a legitimizing device to: (i) educate and inform relevant publics about (actual) changes in their environmental impacts that have been made to conform with changed social expectations, (ii) change perceptions about their environmental initiatives without changing operations, (iii) deflect attention from the issue of concern by highlighting other positive accomplishments, or (iv) seek to change public expectations about reasonable levels of environmental impacts and risk. Thus, if companies suspect a threat to their social and/or political legitimacy, they have an incentive to actively disclose social and environmental information to offset negative impacts that may be detrimental to their reputation and ongoing survival (Clarkson et al., 2008; Deegan, 2002; Patten, 2000, 2002). In the final three of Lindblom's (1993) legitimation strategies, the rhetoric of the social and environmental disclosures may then be different from any underlying reality.

In statistical studies, because legitimacy is not directly observable, it is common for researchers to infer legitimation processes and effects by analysing relationships between observable corporate performance variables, including social and environmental performance and social and environmental reporting (Aerts and Cormier, 2009). For most legitimation strategies, legitimacy theory (at least in the field of sustainability reporting) predicts a negative relationship between social and environmental performance and its reporting (Cho et al., 2012; Clarkson et al., 2011).

The argument for this is that as the expectations evolve of politically

and economically powerful stakeholders regarding corporate social and environmental performance, companies that do not meet these expectations face threats to their legitimacy (Clarkson et al., 2011). That is, companies performing poorly on social and environmental factors face more social and political (or regulatory) pressures and threatened legitimacy (and thus a greater likelihood of costly regulatory actions against them) (Cho et al., 2012). Accordingly, social and environmentally poorly performing companies are more likely to attempt to increase discretionary social and environmental disclosures to change politically and economically powerful stakeholders' perceptions about their actual performance in order to reduce exposures to these social and political pressures (Cho and Patten, 2007).

#### 3.2. Empirical literature and hypothesis development

Several studies in social and environmental reporting research in the field of sustainability accounting have empirically analysed the role of sustainability reporting from the viewpoint of voluntary disclosure and legitimacy theory. Those that found a positive relationship between sustainability performance and reporting include Al-Tuwaijri et al. (2004),<sup>3</sup> Clarkson et al. (2008), and Silva–Gao (2012), whereas those that found a negative relationship include Bewley and Li (2000), Cho et al. (2012), Clarkson et al. (2011), Guidry and Patten (2012), Hughes et al. (2001), and Patten (2002).<sup>4</sup>

Bewley and Li (2000) found results inconsistent with voluntary disclosure theory, although they nevertheless appealed to this theory, while Hughes et al. (2001) concluded that environmental information differed between good, mixed, and poor environmentally performing companies, and that poor environmentally performing companies, and that poor environmentally performing companies made the most disclosures, a finding consistent with legitimacy theory. Patten (2002) also provided results consistent with legitimacy theory, especially for companies in non-environmentally sensitive industries. Clarkson et al. (2008) analysed US companies and found results consistent with voluntary disclosure theory, while Clarkson et al. (2011) in an examination of Australian companies found results consistent with socio-political theories, with both studies employing the same proxy for environmental disclosure. This implies that the application of theories to explain the role of sustainability reporting may differ between countries.

Elsewhere, Guidry and Patten (2012) re-estimated the data used by Cho and Patten (2007) by considering that the control variables such as profitability and leverage used in previous studies had focused on voluntary disclosure theory. However, even after including control variables for voluntary disclosure theory, there was still support for legitimacy theory. Silva–Gao (2012) concluded a positive relationship for US companies between lower greenhouse gas (GHG) emissions and the amount of environmental capital expenditure information. This at least supports the role of voluntary disclosure theory in the relationship between specific environmental performance and reporting.

Studies that have empirically analysed the role of sustainability reporting practices from the perspective of voluntary disclosure theory and legitimacy theory have provided mixed results. Many obtained findings consistent with legitimacy theory rather than voluntary disclosure theory. However, the number of studies whose findings are consistent with voluntary disclosure theory has recently been increasing. Having said that, most have observed the (limited) relationship between specific environmental performance and reporting, which is consistent with the findings in Silva–Gao (2012).

For example, Giannarakis et al. (2016, 2017a, 2017b) revealed various environmental performance indicators such as GHG emissions,

climate change policy, and emission reduction initiatives were positively associated with environmental disclosures. Kang and Gray (2019) found that British multinational companies were less likely to disclose their segment and risk information on a country-by-country basis if they conducted operations in countries with higher levels of country-specific risks. Abubakar Siddique et al. (2021) concluded that companies with better carbon performance were more likely to make more carbon disclosures.

Other than these studies, Acar and Temiz (2020) obtained results consistent with voluntary disclosure theory in an emerging market context and Uyar et al. (2020) found that companies with better CSR performance were more likely to publish one or more CSR reports. Similarly, Lu and Wang (2021) found that companies with better environmental performance disclosed more CSR information. In contrast, Yook et al. (2017) revealed a negative relationship between the disclosed levels of environmental control costs and eco-efficiency performance measures and Luo (2019) found a negative relationship between carbon emission performance and voluntary carbon disclosure. Cong et al. (2020) concluded a positive relationship existed between GHG emissions and the extensiveness of climate change disclosure, thereby supporting legitimacy theory. As a unique study, Li et al. (2017) found that the relationship between environmental performance and environmental disclosures exhibited a U-shaped nonlinear relationship, thereby eliminating the discrepancy between legitimacy and voluntary disclosure theory.

In addition, some studies distinguished between types of disclosure when empirically analysing the role of sustainability reporting. Clarkson et al. (2008) split environmental disclosures into hard disclosures (i.e., governance structure and environmental management systems, environmental performance indicators, and environmental spending) and soft disclosures (i.e., vision and strategy claims, environmental profile, and environmental initiatives) and concluded environmental performance was significantly positive for both hard and soft disclosures. Regardless of the type of disclosures, the findings supported voluntary disclosure theory. In contrast, Hummel and Schlick (2016) classified sustainability disclosures into high quality (in terms of verifiability, reliability, comparability, and consistency) and low quality, and found that companies with better sustainability performance disclosed more high-quality sustainability information whereas those with worse sustainability performance disclosed more low-quality sustainability information. Their findings indicated that voluntary disclosure theory and legitimacy theory simultaneously explained the disclosure quality of information within social and environmental sustainability reporting.

According to those studies with mixed results, companies might potentially have both a legitimation and financial accountability motivation for voluntary corporate reporting, and their reporting philosophy (or motivation) might then depend upon distinct types of reporting practices as Clarkson et al. (2008), Hummel and Schlick (2016), and de Villiers and van Staden (2011) proposed. In other words, they might use different reporting practices to differentiate their reporting philosophy. This is obvious from the findings (in assorted studies) that differing aspects of voluntary reporting support different motivations, such as the volume of content and type of information. However, we cannot clarify the multidimensionality of motivations underlying voluntary corporate reporting if an analysis focuses only on the volume of a report's contents. For that reason, where integrated reporting has appeared as a new form of voluntary corporate reporting for the disclosure of financial and nonfinancial information to investors, companies can refine their reporting philosophy more clearly than before by using integrated reporting and/or social and environmental sustainability reporting (depending on their reporting philosophy).

As suggested, many companies that have adopted integrated reporting have replaced their former publication of a social and environmental sustainability report with an integrated report. However, this does not mean that all companies have switched from social and environmental sustainability reporting to integrated reporting. Even where

 $<sup>^3</sup>$  Although Al-Tuwaijri et al. (2004) do not explicitly refer to voluntary disclosure theory, their argument follows the theory.

<sup>&</sup>lt;sup>4</sup> We omit studies analysing the influence of other factors on sustainability reporting from this review. Zhang and Liu (2020) and da Rocha Garcia et al. (2021) provide useful reviews.

some companies used both types of reporting practices at the same time, the possibility that these companies also had an incentive to refine their voluntary reporting philosophy through different reporting practices did not change. Companies may in fact have multiple motivations. Accordingly, we expect companies with financial accountability motivations to have an incentive to switch from existing social and environmental sustainability reporting to integrated reporting or, for some companies, an incentive to additionally adopt integrated reporting.

However, when analysing the motivations for voluntary corporate reporting in an unfamiliar environment, it is necessary to reflect on the applicability of prior data collection and analysis protocols rather than directly adopting the same methods, as these may not be suited to the different context (such as focusing on the volume of contents in a report). In the context of this study, this is because while voluntary integrated reporting shares certain characteristics with longer-standing social and environmental sustainability reporting, they each have a distinct focus as suggested. Furthermore, the following argument should be also considered.

First, the primary interest in many social and environmental sustainability reporting studies has been the extent (and type and quality) of relevant disclosures in any corporate report. This is because, when engaging in social and environmental sustainability reporting where such reporting has already become usual practice (e.g., Clarkson et al., 2008; Hummel and Schlick, 2016), companies can choose to include greater or lesser amounts of high- or low-quality social and environmental disclosures. In contrast, in the early development stages of integrated reporting (containing concise information), it was a reporting approach (or philosophy) that companies could either adopt or not (Lai et al., 2016). This finds support in the analysis by Eccles et al. (2019) of the sizes of some 50 selected integrated reports in 10 countries (i.e., Brazil, France, Germany, Italy, Japan, Netherlands, South Africa, South Korea, UK, and USA), where they found that the size of integrated reports in South Africa, where integrated reporting is mandatory, was much higher than average in the 10 countries, suggesting that voluntary integrated reporting practices were still developing.

In other work, Pistoni et al. (2018) investigated 116 integrated reports from the IIRC's website and found that integrated reporting quality was low, concluding that there was more attention given to the form of integrated reporting than its content. While some studies have analysed the volume of integrated reporting (e.g., Green and Cheng, 2019; Melloni et al., 2017), there may be sample selection bias in the results because only the practices of proactive companies with already published integrated reports were analysed, in a situation where integrated reporting had not yet become a reporting norm. Studies concerning South Africa, where integrated reporting is mandatory, are exceptional (e.g., Barth et al., 2017).

Second, because the IIRC's integrated reporting framework establishes general principles rather than specific guidelines, there is always flexibility for each company to define the content of their own report, which necessarily reduces the comparability of the disclosed information (Gianfelici et al., 2018). In addition, high-quality integrated reports should be concise and focused and therefore not exceedingly long (Hooks and van Sanden, 2011; International Integrated Reporting Council IIRC, 2013; Melloni et al., 2017), such that integrated reporting only provides "... an entry point to more detailed information outside the designated communication, to which it may be linked" (International Integrated Reporting Council IIRC, 2013).

Therefore, unlike social and environmental sustainability reporting, the volume of disclosures may not capture the essence of integrated reporting.<sup>5</sup> In other words, it is not always the case that the greater the extent of reporting, the higher the quality of reporting. This is reasonable from the background of the development of integrated reporting; while a company's sustainability report can contain considerable information about the company's social and environmental policies, practices, and performance, these reports have become very complex and long in order to meet the information needs of a wide range of stakeholders (Berthelot et al., 2003; Cho et al., 2010; Cormier et al., 2004; de Villiers et al., 2014). They also do not usually present an integrated picture of the interrelationships between financial, social, and environmental performance and risk—when in practice these factors are often highly interrelated (Jensen and Berg, 2012; Maas et al., 2016). Thus, providers of financial capital seeking to incorporate social and environmental factors into their investment decision-making needed to draw upon a range of sources of company-produced information (Epstein and Freedman, 1994; Harte et al., 1991; Rockness and Williams, 1988).

At first impression, examining integrated reporting practices using the extent or volume of disclosed information following the social and environmental and sustainability reporting literature might seem to be more appropriate. However, following such a conventional approach at this early stage of development of integrated reporting risks producing misleading results. Although we respect the methods used in previous social and environmental sustainability reporting studies, a unique way to avoid misleading results is necessary. Therefore, rather than comparing the detailed contents of each company's integrated and sustainability reports, our interest lies in the philosophy of corporate voluntary reporting, which depends on whether and when a company has shifted to the IIRC's integrated reporting or continues with its existing social and environmental sustainability reporting.<sup>6</sup> However, this does not mean our interest contradicts previous studies because some studies such as those of Clarkson et al. (2008) and Hummel and Schlick (2016) are also based on the difference in corporate reporting philosophy, although the focus of reporting practice differs.

From this perspective, and drawing on voluntary disclosure theory, if companies incorporate social and environmental impacts into their strategy and financial decision-making, and the associated financial and nonfinancial impacts are strategically beneficial for investors (Grassmann, 2021), companies with better social and environmental performance will have an incentive to shift to integrated reporting practices as soon as possible. It is important to consider the point of time of the shift in reporting because the sooner companies with a financial accountability motivation inform their investors through integrated reporting, the greater the financial advantage they can expect to obtain from a higher investor valuation.

At the same time, if voluntary reporting is a rhetorical legitimating tool for communicating with a broader range of politically and economically powerful stakeholders, a company with poorer social and environmental performance would have an incentive to continue with social and environmental sustainability reporting practices if possible. In contrast, if integrated reporting is just a developed form of existing self-legitimating sustainability reporting, and the company does not have an incentive to refine its voluntary reporting philosophy through its reporting, a company with poorer social and environmental performance will have an incentive to shift to integrated reporting practices as soon as possible.

The IIRC (2013, p.4) asserts that integrated reporting "... promotes a more cohesive and efficient approach to corporate reporting and aims to improve the quality of information available to providers of financial capital to enable a more efficient and productive allocation of capital".

<sup>&</sup>lt;sup>6</sup> The volume of sustainability report contents often used in social and environmental sustainability reporting research has also been just one proxy to represent actual voluntary reporting practices, and the proxy Y does not always exactly match with actual voluntary reporting practices Y<sup>\*</sup>. For example, we can only describe the relationship between the (visualized) proxy and actual practice as Y = 1 if  $Y^* > 0$  and Y = 0 if  $Y^* \le 0$ , other than  $Y = Y^*$ . Therefore, the selection of a proxy variable is context specific, and includes consideration of reporting quality and philosophy.

<sup>&</sup>lt;sup>5</sup> However, some integrated reports can be quite long.

For companies agreeing with this suggestion—rather than considering integrated reporting to be just a developed form of existing (legitimizing) sustainability reporting—voluntary disclosure theory rather than legitimacy theory should be applicable to integrated reporting. Conversely, "… fundamental to [integrated reporting may be] that it seeks legitimacy in situations where neither the practice nor the (institutional) environment is fixed" (van Bommel, 2014, p.1159). Therefore, we propose the following hypotheses for testing<sup>7</sup>:

**H1a.** Companies with better social and environmental performance are more likely to shift to integrated reporting practices sooner.

H1a also suggests that companies with worse social and environmental performance are more likely to continue with social and environmental sustainability reporting practices longer.

**H1b.** Companies with worse social and environmental performance are more likely to shift to integrated reporting practices sooner.

#### 4. Research design

#### 4.1. Regression model

We employ *survival analysis* to assess these hypotheses because when companies adopt integrated reporting, they subsequently continue with the practice. Survival analysis focuses on the hazard rate, which is the probability of timing and duration for a company to shift to integrated reporting where social and environmental sustainability reporting has been its previous practice. This enables us to include time dimensions regarding the shift to (or initial adoption of) integrated reporting practices. In other words, because survival analysis analyses the time to event (the length of time before each company in the sample initially publishes an integrated report), we can clarify which companies are *early* adopters of integrated reporting. This contrasts with the ordinary least squares (OLS) and probit models often used in accounting research which focus on the continuation of reporting practices within annual and sustainability reports.

Suppose the random variable *T* represents the duration interval *until* a company initiates integrated reporting, which has a continuous probability distribution f(t), where *t* is a realization of *T*. Then, the cumulative distribution function of the initiation of integrated reporting is as follows:

$$F(t) = Pr(T \le t) = \int_{0}^{t} f(u) du$$

The survival function S(t) is the probability that the duration interval (until a company initiates integrated reporting) is at least of length t.

$$S(t) = Pr(T > t) = 1 - F(t)$$

where  $0 \le S(t) \le 1$ , S(0) = 1,  $\lim_{t\to\infty} S(t) = S(\infty) = 0$ , and the hazard function h(t) is the ratio of the continuous probability distribution f(t) and the survival function S(t). This measures the probability of timing and duration for companies that have not yet adopted integrated reporting practices but that will initially adopt them in the following

period 
$$\left( = \lim_{\Delta t \to 0} \frac{Pr(t \le T < t + \Delta t | t \le T)}{\Delta t} \right)$$
  
 $h(t) = \frac{f(t)}{S(t)}$ 

Within survival analysis, the proportional hazards model estimating the parameter(s) without requiring estimation of the baseline hazard is a popular regression method for analysing the effect of explanatory variables on the hazard rate (Cox, 1972; Greene, 2003).

$$h_{it} = h_0(t) exp(\beta_1 SEP_{it} + \beta_2 CONT_{it})$$

where *h* is the hazard rate,  $h_0(t)$  is the baseline hazard, *SEP* is a variable that captures social and environmental performance, *CONT* is a control variable, and  $\beta$  are the estimated coefficients.

#### 4.2. Sample selection

The data used for the regression analyses are pooled data for the 100largest companies listed on each of the Tokyo Stock Exchange and London Stock Exchange (FTSE 100 and their equivalent Tokyo Stock Exchange companies) from 2012 (just after the IIRC introduced its basic principles for the integrated reporting framework) until 2017. The FTSE 100 companies is a common sample for London Stock Exchange listed companies, as is the analysis of the 100-largest companies on the Tokyo Stock Exchange. However, because the sampling is not random, there can be bias in regression estimations, a situation known as sample selection bias (Heckman, 1979). Because our sample consists only of the very largest companies, we cannot deny the potential for sample selection bias in our data and the possibility that our findings will not then be generalizable. To correct for the latent sample selection bias, we employ the Heckman inverse Mills ratio method as detailed later.

We exclude financial sector companies from the sample because they differ from other companies in other sectors in the form and content of their financial reporting. The total number of company-year observations across the six years 2012–2017 in our unbalanced panel of data is 426 for 82 Tokyo listed nonfinancial companies and 320 observations for 71 London listed nonfinancial companies.

#### 4.3. Variable measurement

The dependent variable is the *hazard rate* for the period in years from 2012 until the date a company first publishes an integrated report that claims that it follows the IIRC's definition of integrated reporting. The independent variable used to capture a company's social and environmental performance is the social and environmental score. We alternatively specify the emission reduction score as a robustness check.

The control variables are company size as measured by the logarithm of net sales (in pounds sterling) (e.g., Al-Tuwaijri et al., 2004; Bewley and Li, 2000; Cho and Patten, 2007; Cho et al., 2012; Clarkson et al., 2008, 2011; Nishitani et al., 2020; Patten, 2000, 2002), profitability as measured by the return on assets (ROA) (e.g., Al-Tuwaijri et al., 2004; Bewley and Li, 2000; Cho et al., 2012; Clarkson et al., 2008, 2011; Nishitani et al., 2020), financial leverage as measured by the debt ratio (e.g., Cho et al., 2012; Clarkson et al., 2008, 2011; Nishitani et al., 2020), and industry-specific characteristics as measured by industry sector dummies (e.g., Bewley and Li, 2000; Cho et al., 2012; Clarkson et al., 2008, 2011; Nishitani et al., 2020; Patten, 1991, 2002). While these controls either match or exceed the minimum required control variables commonly employed in previous studies in the area, Guidry and Patten (2012) noted that including many control variables unnecessarily would not significantly change the results.

Table 1 lists the definitions of these dependent and independent variables, Table 2 provides their descriptive statistics, and Table 3 presents the correlation matrix for the variables. To assess the potential for harmful multicollinearity, we use the pairwise correlation coefficients between the independent variables, revealing the highest correlations to be 0.538 for the Tokyo listed companies and 0.542 for the London listed companies, both between the social and environmental score and company size. While the correlations between the social and environmental score and the emission reduction score are higher, 0.898 for the

<sup>&</sup>lt;sup>7</sup> Given that we empirically evaluate these hypotheses using data for the largest companies listed on each of the Tokyo and London stock exchanges, we correct for the latent sample selection bias associated with sampling only the largest companies. Were this not the case, the hypotheses should commence with "Large companies with …" rather than "Companies with …".

Variable definitions and sources.

Definition	Database
Hazard rate: the hazard rate is the period in years from 2012 until the date a company first publishes an integrated report claiming to follow the IIRC's definitions of integrated reporting. Calculated using data on whether and when a company adopts integrated reporting (IR dummy), omitting companies that have initiated integrated reporting by the previous year from the sample in this and any subsequent year.	Hand-collected from the IIRC's website and each company's webpage
<ul> <li>Social and environmental score: the average of the environmental and social scores in the ASSET4 database provided by Thomson Reuters. Social and environmental sustainability information in an integrated report and the sustainability report covers not only environmental but also social aspects. Thus, social and environmental performance should be simultaneously considered for the analyses. However, if included in the same regression (as proxies for social and environmental performance, respectively), they will cause a multicollinearity problem because they tend to be highly correlated. To avoid this problem, we employ the average of the social and environmental scores as a single variable. According to ASSET4's definitions, the environmental score measures a company's impact on living and non-living natural systems, including air, land, and water, as well as complete ecosystems. It thus reflects how well a company uses best management practices to reduce environmental risks and capitalize on environmental opportunities. In contrast, ASSET4's social score measures a company's capacity to generate trust and loyalty with its workforce, customers, and society, through its use of best management practices. According to ASSET4's definitions, this reflects the company's reputation and the health of its social licence to operate. Note that the estimation results do not change majorly even if we employ both the social and environmental commitment and effectiveness towards reducing environmental emissions in its production and operational processes. It reflects a company's capacity to reduce air emissions (greenhouse gases, F-gases, ozone-depleting substances, NOx, and SOx, etc.), waste, hazardous waste, water discharges, spills or its impacts on biodiversity, and to form partnerships with environmental organizations to reduce the environmental impact of the company in the local or broader community.</li> </ul>	Datastream database
<ul> <li>Company size: the logarithm of net sales (in pounds sterling).</li> <li>Return on assets (ROA): net income divided by total assets.</li> <li>Debt ratio: debt divided by equity.</li> <li>Industry sector dummies: dummy variables that take a value of one if a company belongs to the mining, construction &amp; real estate, manufacturing with larger environmental impacts, manufacturing with smaller environmental impacts, utility (electric power &amp; gas), transportation, information &amp; communication, or sales &amp; service (wholesale, retail trades &amp; service) sectors, respectively. Because industry-specific influences go beyond industry-specific as environmental impacts, this series of industry sector dummies is preferable to dummy variables simply classified as environmentally sensitive or socially exposed industries. However, (at least) whether social and environmental impacts of companies in the manufacturing industry are larger or smaller depends on their sub-classification within this industry sector. For example, Patten (1991, 2002) categorizes chemical, metals, paper, and petroleum industries as industries with larger environmental impacts. Accordingly, we divide companies in the manufacturing sector in our sample into those in manufacturing with larger environmental impacts and those</li> </ul>	Bloomberg database

in manufacturing with smaller environmental impacts based on this categorization.

Tokyo listed companies and 0.768 for the London listed companies, we do not include both variables in the same regression. We also calculate the variance inflation factors (VIFs) for both sets of companies (results not shown), with the highest VIFs being 2.03 and 2.11 for the Tokyo and London listed companies, respectively. Both the correlation analysis and the VIFs suggest we need not be too concerned about the potential for harmful multicollinearity.

#### 5. Estimation results

Tables 4 and 5 provide the estimation results concerning the influence of social and environmental performance on the probability of timing and duration of a shift to integrated reporting practices (or hazard rate) for our sample of Tokyo and London listed companies. The regressions in Table 4 assume all companies are homogeneous, while Table 5 assumes that they are heterogeneous in terms of industry sector when capturing the influence of social and environmental performance.

Each table details the results for Models (1T), (2T), (1L), and (2L) and for Panels A and B. Models (1T) and (2T) and Models (1L) and (2L) provide the estimation results for Tokyo (T) and London (L) listed companies for the periods 2012–2014 and 2012–2017, respectively. We use these different periods to identify the dynamics of the motivations for companies shifting to integrated reporting or continuing with social and environmental sustainability reporting. Panels A and B provide estimation results with/without taking the latent sample selection bias into consideration, because our sampling is not random (i.e., the largest-100 companies). For this purpose, all models in Panel B additionally include the inverse Mills ratio as a correction term following Miller et al. (2008).

The inverse Mills ratio is from a probit regression estimating the likelihood on becoming a largest-100 company using all companies listed on the Tokyo and London stock exchanges without missing values over the 6 years of our sample period (2012-2017) (17,698 observations for 3210 Tokyo listed companies and 16,826 observations for 3531 London listed companies). In the probit regression, we use the capital intensity ratio (i.e., the book value of tangible fixed assets divided by the number of employees) as a variable potentially influencing the likelihood of becoming a largest-100 company but not directly influencing the adoption of integrated reporting in addition to the control variables. This is reasonable as the managerial finance literature is undecided about whether capital intensity influences a company's firm value positively or negatively (e.g., Bourke et al., 2020; Lee and Xiao, 2011; Pattanayak, 2009). Indeed, we find capital intensity is a significantly positive influence on becoming one of the 100-largest Tokyo listed companies, but a significantly negative influence for London listed companies (results not shown). However, it is unlikely that capital intensity directly influences a company's voluntary reporting philosophy at present. The following subsections present the overall and industry sector-level estimation results and the discussion.

#### 5.1. Overall estimation

We first examine the estimation results in Table 4. All models include seven industry sector dummies to control for industry-specific influences, although we do not display their coefficients (the same applies to Tables 5–7).

Models (1T) and (2T) in Panels A and B show that the social and environmental score is significantly positive. Although the results in

#### K. Nishitani et al.

#### Table 2

Descriptive statistics.

Tokyo listed companies	Obs.	Mean	SD	Min	Max	Skewness	Kurtosis
IR dummy	426	0.056	0.231	0	1	3.848	15.810
Social and environmental score	426	73.587	24.363	7.935	95.725	-1.433	3.753
Emission reduction score	424	78.157	24.751	9.120	96.300	-1.676	4.596
Company size	426	23.079	1.084	20.577	25.864	0.054	2.566
ROA	426	5.003	4.250	-12.572	31.985	1.067	7.988
Debt ratio	426	64.890	79.928	0	455.559	2.204	8.625
Industry dummy							
Mining	426	0.014	0.118	0	1	8.247	69.014
Construction & real estate	426	0.052	0.222	0	1	4.052	17.418
Manufacturing with larger environmental impacts	426	0.092	0.289	0	1	2.833	9.024
Manufacturing with smaller environmental impacts	426	0.521	0.500	0	1	-0.085	1.007
Utility	426	0.028	0.166	0	1	5.703	33.529
Transportation	426	0.056	0.231	0	1	3.848	15.810
Information & communication	426	0.087	0.282	0	1	2.934	9.609
Sales & service	426	0.150	0.358	0	1	1.958	4.833
London listed companies	Obs.	Mean	SD	Min	Max	Skewness	Kurtosis
IR dummy	320	0.069	0.253	0	1	3.409	12.619
Social and environmental score	320	78.216	17.066	9.415	95.590	-1.764	5.924
Emission reduction score	319	79.801	18.201	15.960	96.250	-1.835	5.958
Company size	320	22.322	1.573	18.446	26.460	0.196	3.070
ROA	320	10.920	26.562	-18.899	235.464	6.987	55.548
Debt ratio	320	82.498	85.841	0	495.231	2.226	8.891
Industry dummy							
Mining	320	0.028	0.166	0	1	5.708	33.584
Construction & real estate	320	0.109	0.313	0	1	2.503	7.266
Manufacturing with larger environmental impacts	320	0.147	0.355	0	1	1.995	4.981
Manufacturing with smaller environmental impacts	320	0.234	0.424	0	1	1.254	2.573
Utility	320	0.072	0.259	0	1	3.315	11.990
Transportation	320	0.019	0.136	0	1	7.096	51.352
Information & communication	320	0.122	0.328	0	1	2.312	6.344
Sales & service	320	0.269	0.444	0	1	1.043	2.088

Panel B show that the estimate for the inverse Mills ratio is significantly negative, the results in Panels A and B do not differ, meaning that sample selection bias is not a significant problem when we assume all Tokyo listed companies are homogeneous. These results suggest that Tokyo listed companies with better social and environmental performance are more likely to have shifted to integrated reporting practices early and by 2014 at the latest, which supports Hypothesis 1a for Tokyo listed companies.

In contrast, Models (1L) and (2L) in Panels A and B show that the social and environmental score does not have a statistically significant influence on the shift to integrated reporting for London listed companies. As with the estimates for the Tokyo listed companies, the estimation results in Panels A and B do not differ, meaning that sample selection bias is also not a significant problem for London listed companies. These results suggest that London listed companies with better or worse social and environmental performance are not more likely to have shifted to integrated reporting practices by 2017, which fails to support either Hypothesis 1a or Hypothesis 1b for London listed companies in either period.

#### 5.2. Industry sector-level estimation

We next examine the estimation results in Table 5. To analyse the influence of social and environmental performance on the timing and duration of a shift to integrated reporting practices in different industry sectors, the models additionally include the interaction terms between the social and environmental score and the eight industry sector dummies.

Model (1T) in Panels A and B shows that the interaction terms between the social and environmental score and an industry sector dummy are significantly positive for manufacturing with larger environmental impacts, manufacturing with smaller environmental impacts, and the information & communication sectors, and significantly negative for the mining sector. Model (2T) in Panel A shows that the interaction terms for the social and environmental score and the industry sector dummies are only significantly positive for the manufacturing with smaller environmental impacts sector. However, Model (2T) in Panel B shows that the interaction terms for the social and environmental score and industry sector dummy are additionally significantly negative for the mining and utility sectors. Although the estimation results for Model (1T) in Panels A and B do not differ, those for Model (2T) in Panels A and B differ, suggesting the underestimation of the (negative) influence of the social and environmental score is because of sample selection bias for Model (2T) in Panel A.

These results suggest that Tokyo listed companies with better social and environmental performance in the manufacturing with smaller environmental impacts sector, and those with worse social and environmental performance in the mining sector are more likely to have shifted to integrated reporting practices by 2014 at the latest, whereas those with worse social and environmental performance in the utility sector are more likely to have shifted to integrated reporting practices by 2017 at the latest. This supports Hypothesis 1a for Tokyo listed companies in the manufacturing with smaller environmental impacts sector, and Hypothesis 1b for those in the mining and utility sectors. However, while Tokyo listed companies with better social and environmental performance in the manufacturing with larger environmental impacts and information & communication sectors are more likely to have shifted to integrated reporting practices by 2014 at the latest, these findings are not robust over the longer period (by 2017).

Model (1L) in Panel A shows that the interaction term between the social and environmental score and the industry sector dummy is significantly positive for the manufacturing with larger environmental impacts, manufacturing with smaller environmental impacts, and the sales & service sectors, and significantly negative for the transportation sector for London listed companies. However, Model (1L) in Panel B shows that the interaction term between the social and environmental score and the industry sector dummy is not significant for companies in the transportation sector, implying the overestimation of the (negative)

Table 3 Correlation matrix.

Tokyo listed companies		[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]	[10]	[11]	[12]	[13]	[14]
[1]	IR dummy	1.000													
[2]	Social and environmental score	0.109	1.000												
[3]	Emission reduction score	0.125	0.898	1.000											
[4]	Company size	0.088	0.538	0.519	1.000										
[5]	ROA	-0.067	-0.397	-0.364	-0.529	1.000									
[6]	Debt ratio	-0.020	-0.042	-0.073	0.341	-0.453	1.000								
[7]	Mining	-0.029	0.017	0.082	-0.037	-0.025	-0.070	1.000							
[8]	Construction & real estate	-0.011	-0.123	-0.141	-0.047	-0.171	0.361	-0.028	1.000						
[9]	Manufacturing with larger environmental impacts	0.028	0.135	0.101	-0.079	0.057	-0.162	-0.038	-0.074	1.000					
[10]	Manufacturing with smaller environmental impacts	-0.031	0.144	0.167	-0.033	0.068	-0.315	-0.125	-0.243	-0.331	1.000				
[11]	Utility	-0.042	0.026	0.008	0.066	-0.129	0.268	-0.020	-0.040	-0.054	-0.178	1.000			
[12]	Transportation	-0.060	-0.092	0.023	0.019	-0.139	0.247	-0.029	-0.057	-0.078	-0.255	-0.042	1.000		
[13]	Information & communication	-0.003	-0.117	-0.190	0.035	0.146	-0.004	-0.037	-0.072	-0.098	-0.322	-0.053	-0.075	1.000	
[14]	Sales & service	0.097	-0.100	-0.124	0.082	0.007	0.090	-0.050	-0.098	-0.134	-0.439	-0.072	-0.103	-0.130	1.000
London listed companies		[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]	[10]	[11]	[12]	[13]	[14]
[1]	IR dummy	1.000													
[2]	Social and environmental score	0.072	1.000												
[3]	Emission reduction score	0.065	0.768	1.000											
[4]	Company size	0.127	0.542	0.468	1.000										
[5]	ROA	-0.018	-0.313	-0.384	-0.354	1.000									
[6]	Debt ratio	-0.064	0.162	0.018	0.103	-0.139	1.000								
[7]	Mining	0.327	0.019	0.051	0.032	0.005	-0.115	1.000							
[8]	Construction & real estate	-0.056	-0.003	0.060	-0.308	-0.040	-0.249	-0.060	1.000						
[9]	Manufacturing with larger environmental impacts	-0.043	0.203	0.196	0.320	-0.086	-0.069	-0.071	-0.145	1.000					
[10]	Manufacturing with smaller environmental impacts	-0.005	0.047	0.010	-0.049	-0.034	-0.043	-0.094	-0.194	-0.230	1.000				
[11]	Utility	-0.028	0.075	-0.029	0.125	-0.051	0.310	-0.047	-0.098	-0.116	-0.154	1.000			
[12]	Transportation	-0.038	-0.147	0.063	-0.013	-0.013	-0.069	-0.024	-0.048	-0.057	-0.077	-0.039	1.000		
[13]	Information & communication	0.050	-0.134	-0.187	-0.187	-0.041	-0.030	-0.063	-0.131	-0.155	-0.206	-0.104	-0.052	1.000	
[14]	Sales & service	-0.053	-0.112	-0.093	0.065	0.191	0.177	-0.103	-0.212	-0.252	-0.335	-0.169	-0.084	-0.226	1.000

Influence of social and environmental performance on hazard rate for integrated reporting.

	(1T) 2012–2014		(2T)		(1L)		(2L)		
			2012-2017		2012-2014		2012-2017		
Panel A	Coefficient	Robust S.E.	Coefficient	Robust S.E.	Coefficient	Robust S.E.	Coefficient	Robust S.E.	
Social and environmental score	0.070	0.022***	0.033	0.017*	0.047	0.029	0.019	0.023	
Company size	-0.351	0.273	0.017	0.236	0.194	0.183	0.289	0.158*	
ROA	-0.023	0.089	-0.054	0.048	0.012	0.013	0.003	0.008	
Debt ratio	0.0002	0.003	-0.003	0.003	-0.002	0.004	-0.002	0.003	
Industry sector dummies	у	es	у	es	у	es	yes		
Wald test (p-value)	0.0	000	0.0	0.003		000	0.000		
Log pseudolikelihood	-42.898		-94.323		-64.422		-81.030		
Observations	232		426		176		320		
	(1T)		(2T) 2012–2017		(1L)		(2L)		
	2012-2014				2012-2014		2012–2017		
Panel B	Coefficient	Robust S.E.	Coefficient	Robust S.E.	Coefficient	Robust S.E.	Coefficient	Robust S.E.	
Social and environmental score	0.081	0.025***	0.033	0.019*	0.040	0.027	0.026	0.023	
Company size	-2.173	0.867**	-0.873	0.553	-0.048	0.419	0.528	0.441	
ROA	-0.088	0.102	-0.086	0.052*	-0.0001 0.016		0.011	0.017	
Debt ratio	0.005	0.004	-0.002	0.004	-0.002	0.005	-0.002	0.003	
Inverse Mills ratio	-2.947	1.535*	-1.615	0.950*	-1.284	1.915	1.223	1.729	
Industry sector dummies	у	es	yes		у	es	yes		
Wald test (p-value)	0.0	000	0.002		0.0	000	0.000		
Log pseudolikelihood	-41	.642	-92.669		-64	.257	-74.952		
Observations	2	28	417		1	69	304		

\*\*\*, \*\*, and \* indicate that the coefficient is significantly different from zero at the 1%, 5%, and 10% levels, respectively.

influence of the social and environmental score due to sample selection bias for Model (1L) in Panel A. Conversely, Model (2L) in Panels A and B indicates that the interaction term between the social and environmental score and the industry sector dummy is significantly positive for the manufacturing with smaller environmental impacts sector.

These results suggest that London listed companies with better social and environmental performance in manufacturing with smaller environmental impacts are more likely to have shifted to integrated reporting practices by 2014 at the latest, which supports Hypothesis 1a for London listed companies in the manufacturing with smaller environmental impacts sector. Alternatively, while London listed companies with better social and environmental performance in manufacturing with larger environmental impacts and the sales & service sectors are more likely to have shifted to integrated reporting practices by 2014 at the latest, these findings are not robust over the longer period (by 2017).

#### 5.3. Robustness check

In addition to the social and environmental score, we also specify ASSET4's emission reduction score as a proxy for each company's social and environmental performance for a robustness check. This proxy captures the characteristics of specific social and environmental performance. Tables 6 and 7 provide the estimation results for the overall and industry sector-level estimation, respectively (Panel B only).

As shown in Table 6, while Models (1T) and (2T) show that the emission reduction score is significantly positive, Models (1L) and (2L) do not. Thus, the estimation results for both the Tokyo and London listed companies are like those in Table 4, which robustly supports Hypothesis 1a for Tokyo listed companies.

In Table 7, the estimates for Models (1T) and/or (2T) show that the interaction term between the emission reduction score and the industry sector dummy is significantly positive in the manufacturing with smaller environmental impacts, information & communication, and sales & service sectors, and significantly negative in the mining sector. In addition, Models (1L) and/or (2L) show that the interaction terms between the emission reduction score and the industry sector dummy are significantly positive in the mining, utility, and sales & service sectors, and significantly negative in the information & communication sector.

However, this negative influence for the information & communication sector weakens over time (by 2017).

Thus, while there is some similarity between these results and those in Table 5, they differ in terms of demonstrating the positive influence of the emission reduction score over the social and environmental score in more industry sectors. Indeed, Models (1T) and (2T) support Hypothesis 1a for Tokyo listed companies in the construction & real estate, information & communication, and sales & service sectors as well as the manufacturing with smaller environmental impacts sector. Furthermore, Models (1L) and (2L) also support Hypothesis 1a for London listed companies in the mining, utility, and sales & service sectors, but not those in the manufacturing with smaller environmental impacts sector.

These results imply that Hypothesis 1a is sufficiently robust, and even stronger when social and environmental performance is proxied by more specific environmentally oriented variables. This is consistent with previous findings in the literature. However, Hypothesis 1b is not very robust because Model (2T) supports Hypothesis 1b only for Tokyo listed companies in the mining sector (before 2017).

#### 6. Discussion

An increasing number of providers of financial capital are recognizing long-term financial risks and impacts arising from corporate social and environmental policies, practices, and outcomes. This has led to recognition of the need to integrate relevant social and environmental sustainability information with the financial reporting in corporate reports targeted at investors. Policymakers have responded to this need by developing integrated reporting, with the IIRC leading development of an integrated reporting framework. Motivations for managers to shift or additionally adopt this new form of voluntary corporate reporting, combining elements of existing forms of voluntary social and environmental sustainability reporting with a much narrower focus on reporting to investors, could vary from the legitimation motivations that the academic literature has found to underlie broader social and environmental sustainability reporting. In this context, the aim of this study was to ascertain the apparent primary motivation for corporations to shift to integrated reporting or to continue with social and environmental sustainability reporting. To address these aims, we hand collected data and

Influence of social and environmental performance on hazard rate of integrated reporting at the industry sector level.

	(1T)		(2T)		(1L)		(2L)		
	2012-2014		2012-2017		2012-2014		2012-2017		
Panel A	Coefficient	Robust S.E.							
Social and environmental score									
×Mining	-0.063	0.033*	-0.012	0.034	-0.080	0.057	-0.044	0.038	
×Construction & real estate	0.004	0.021	0.001	0.024	-0.066	0.071	-0.027	0.030	
$\times$ Manufacturing with larger environmental impacts	0.545	0.229**	0.020	0.024	0.643	0.200***	0.238	0.172	
$\times$ Manufacturing with smaller environmental impacts	0.070	0.025***	0.065	0.026**	0.145	0.054***	0.145	0.056***	
×Utility	-0.007	0.102	-0.033	0.096	-0.047	0.048	-0.043	0.044	
×Transportation	0.002	0.021	-0.001	0.021	-0.179	0.096*	-0.048	0.094	
×Information & communication	0.043	0.018**	0.040	0.025	-0.009	0.022	-0.005	0.022	
×Sales & service	0.071	0.047	0.025	0.025	0.524	0.169***	0.011	0.039	
Company size	-0.284	0.254	0.019	0.231	0.452	0.207**	0.382	0.131***	
ROA	-0.002	0.080	-0.059	0.049	0.005	0.018	0.002	0.009	
Debt ratio	0.0001	0.003	-0.004	0.003	-0.003	0.005	-0.003	0.004	
Industry sector dummies	у	es	yes		У	es	yes		
Wald test (p-value)	0.0	000	0.000		0.000		0.000		
Log pseudolikelihood	-42	2.241	-93.742		-58.760		-78	3.449	
Observations	2	32	426		176		320		
	(1T)		(2T)		(1L)		(2L)		
	2012-2014		2012–2017		2012–2014		2012–2017		
Panel B	Coefficient	Robust S.E.							
Social and environmental score									
×Mining	-0.101	0.034***	-0.081	0.031***	-0.072	0.056	-0.044	0.042	
$\times$ Construction & real estate	0.002	0.023	-0.002	0.026	0.005	0.066	0.008	0.217	
$\times$ Manufacturing with larger environmental impacts	0.569	0.235**	0.002	0.033	0.485	0.125***	0.265	0.192	
$\times$ Manufacturing with smaller environmental impacts	0.072	0.025***	0.062	0.024***	0.144	0.063**	0.151	0.053***	
×Utility	0.025	0.032	-0.096	0.033***	-0.089	0.058	-0.025	0.052	
× Transportation	0.003	0.021	-0.031	0.039	-0.002	0.151	-0.028	0.328	
$\times$ Information & communication	0.085	0.033***	0.073	0.047	-0.024	0.023	0.001	0.022	
×Sales & service	0.077	0.057	0.019	0.027	0.531	0.176***	0.015	0.037	
Company size	-2.314	0.892***	-0.963	0.536*	-0.029	0.408	0.568	0.387	
ROA	-0.071	0.097	-0.088	0.052*	-0.019	0.024	0.009	0.020	
Debt ratio	0.006	0.004	-0.001	0.004	-0.004	0.006	-0.003	0.004	
Inverse Mills ratio	-3.336	1.577**	-1.793	0.991*	-2.561	2.020	0.986	1.785	
Industry sector dummies	у	es	yes		у	es	yes		
Wald test (p-value)	0.0	000	0.000		0.0	000	0.0	000	
Log pseudolikelihood	-40	).877	-91	.853	-58	3.328	-72	2.557	
Observations	2	28	4	17	1	69	304		

\*\*\*, \*\*, and \* indicate that the coefficient is significantly different from zero at the 1%, 5%, and 10% levels, respectively.

#### Table 6

Robustness check for the influence of social and environmental performance on the hazard rate for integrated reporting.

	(1T) (2T)			(1L)		(2L)			
	2012-2014		2012-2017		2012-2014		2012-2017		
Panel B	Coefficient	Robust S.E.							
Emission reduction score	0.082	0.034**	0.073	0.025***	0.009	0.022	0.020	0.024	
Company size	-1.471	0.820*	-0.775	0.549	0.002	0.396	0.585	0.424	
ROA	-0.089	0.095	-0.070	0.057	-0.008	0.019	0.013	0.020	
Debt ratio	0.004	0.004	0.001	0.004	-0.002	0.005	-0.001	0.003	
Inverse Mills ratio	-1.694	1.524	-1.177	1.039	-1.819	1.899	1.152	1.846	
Industry sector dummies	ye	es	у	yes		es	yes		
Wald test (p-value)	0.0	0.010		0.000		000	0.000		
Log pseudolikelihood	-41	-41.924		-90.330		.990	-74.983		
Observations	23	32	4	19	1	70	306		

\*\*\*, \*\*, and \* indicate that the coefficient is significantly different from zero at the 1%, 5%, and 10% levels, respectively.

undertook a survival analysis of the probability of timing and the duration of a shift to integrated reporting by companies listed on the world's two largest stock exchanges outside the US, namely, Tokyo and London.

In the overall estimation, the data demonstrate a general trend towards accountability motivations consistent with voluntary disclosure theory for Tokyo listed companies, which supports Hypothesis 1a, but that there is no support for either Hypothesis 1a or Hypothesis 1b for London listed companies. The robustness check results also indicate the same trend. Therefore, Tokyo listed companies appear to regard integrated reporting as a tool for transparency and accountability to enhance shareholder value by providing information to aid improved decision-making by their investors, consistent with the IIRC's stated aims for integrated reporting. At the same time, they regard social and environmental sustainability reporting as a rhetorical legitimating tool for a broader range of politically and economically powerful

Robustness check for the influence of social and environmental performance on the hazard rate for integrated reporting at the industry sector level.

	(1T)		(2T)		(1L)		(2L)		
	2012-2014		2012-2017		2012–2014		2012-2017		
Panel B	Coefficient	Robust S.E.							
Emission reduction score									
×Mining	0.020	0.154	-0.139	0.034***	0.029	0.031	0.050	0.024**	
×Construction & real estate	0.014	0.018	0.110	0.052**	0.006	0.073	-0.021	0.087	
$\times$ Manufacturing with larger environmental impacts	-0.017	0.041	-0.008	0.031	-0.032	0.054	0.057	0.076	
×Manufacturing with smaller environmental impacts	0.352	0.208*	0.121	0.057**	0.017	0.042	0.019	0.047	
×Utility	0.067	0.085	-0.039	0.040	5.016	4.656	0.543	0.252**	
× Transportation	-0.005	0.034	-0.028	0.040	0.016	0.079	0.020	0.079	
×Information & communication	0.273	0.165*	0.126	0.066*	-0.050	0.026*	-0.028	0.026	
×Sales & service	0.019	0.028	0.064	0.033*	0.269	0.147*	0.213	0.087**	
Company size	-2.314	1.137**	-1.054	0.600*	-0.119	0.456	0.675	0.540	
ROA	-0.180	0.121	-0.093	0.061	0.016	0.027	0.026	0.028	
Debt ratio	0.006	0.005	0.001	0.004	-0.001	0.005	-0.0005	0.004	
Inverse Mills ratio	-2.959	2.007	-1.664	1.162	-3.143	2.127	1.038	2.305	
Industry sector dummies	yes		yes		yes		yes		
Wald test (p-value)	0.000		0.000		0.000		0.000		
Log pseudolikelihood	-40	.198	-89.405		-60.852		-71.826		
Observations	23	32	4	419		170		06	

\*\*\*, \*\*, and \* indicate that the coefficient is significantly different from zero at the 1%, 5%, and 10% levels, respectively.

stakeholders. These findings suggest that Tokyo listed companies can refine their voluntary reporting philosophy using their reporting practices.

Earlier sustainability reporting studies indicate that it is possible that Japanese companies may intrinsically have a predisposition toward transparency and accountability as anticipated by voluntary disclosure theory (e.g., Kokubu et al., 2012). Kokubu et al. (2012) analysed the factors underlying the diffusion of sustainability reporting from the perspective of stakeholder theory and revealed the influence of investors on sustainability reporting practices in Japanese companies.<sup>8</sup> By comparing their results with those of previous studies on Japanese companies, Kokubu et al. (2012) suggested the possibility that the influential stakeholders for sustainability reporting practices in Japanese companies had shifted from final consumers to investors. This implied that environmental information had become important in the same way as financial information for investment decisions.<sup>9</sup>

Because nearly all Japanese listed companies have implemented sustainability reporting practices using stand-alone sustainability reports (whose role is to meet the information needs of a wide range of stakeholders), sustainability reporting has a limitation in displaying a company's future actions and plans specifically for economic value creation to providers of financial capital. Therefore, it is possible that Japanese companies with financial accountability motivations consider that integrated reporting is more appropriate than social and environmental sustainability reporting for this purpose. Because social and environmental sustainability reporting in annual reports has never been common in Japanese companies, such a tendency may be particularly stronger.

In contrast, there is no consensus in the analysis of our sample of

London listed companies concerning the accountability or legitimacy motivation and expected role for integrated reporting. However, the proportion of the largest London listed companies that have adopted integrated reporting practices during the period covered by this study is higher than the proportion of the Tokyo listed companies that did so. Although data for some of our sample of London listed companies are consistent with an accountability motivation to shift to integrated reporting, the statistical influence of these companies in our overall sample may then cancel out through the opposing influence. Accordingly, because the robustness check supports an accountability motivation in many but not all industries, it is possible that some London listed companies display an accountability motivation in their shift to integrated reporting.

In the industry sector-level estimation, the data demonstrate a trend towards accountability motivations consistent with voluntary disclosure theory for Tokyo and London listed companies in the manufacturing with smaller environmental impacts sector, and a trend towards legitimating motivations consistent with legitimacy theory for Tokyo listed companies in the mining and utility sectors. In addition, the data also demonstrate a trend towards accountability motivations for Tokyo and London listed companies in other sectors. However, because these trends for companies in other sectors weaken over time, their motivations are not robust over the long term (from 2012 to 2017). That is, as time passes, the motivations to shift to integrated reporting or to continue with social and environmental sustainability reporting based on the underlying corporate reporting philosophy of financial accountability and/or rhetorical legitimation will become fixed.

Because the manufacturing with smaller environmental impacts sector is, if anything, among the potentially less environmentally harmful industries, and the mining and utility sectors are generally regarded as potentially more environmentally harmful industries, these estimation results support Hypothesis 1a for Tokyo and London listed companies in potentially less environmentally harmful industries, and Hypothesis 1b for Tokyo listed companies in potentially more environmentally harmful industries. Thus, the influence of social and environmental performance on the probability of the timing and duration of a shift to integrated reporting practices appears to differ between industries among Tokyo and London listed companies.

On the one hand, Tokyo and London listed companies in potentially less environmentally harmful industries appear motivated to shift to integrated reporting as a transparency and accountability tool to enhance shareholder value by providing reliable information to aid improved decision-making by their investors. At the same time, they

<sup>&</sup>lt;sup>8</sup> In contrast, Liesen et al. (2015) analysed the factors affecting the diffusion of GHG emission reporting in European companies from the perspective of stakeholder theory and found no evidence that investors influenced GHG emissions disclosure.

<sup>&</sup>lt;sup>9</sup> Given this, assurance practices to enhance the credibility of social and environmental information is another topic to be clarified (e.g., Ackers and Eccles, 2015; Haider and Kokubu, 2015). However, because the (third-party) assurance of social and environmental reporting currently provides assurance of only *partially* reported data and claims, we cannot really say that the quality of the social and environmental reporting with assurance is higher. Indeed, Nishitani et al. (2020) found that the stock market did not always positively evaluate such assured reporting.

appear motivated to continue with social and environmental sustainability reporting as a rhetorical legitimating tool for a broader range of politically and economically powerful stakeholders. These companies may then find it less challenging to incorporate social and environmental factors into their strategies because they have a relatively long history of implementing comprehensive sustainability management (Nishitani, 2011). Therefore, it is possible that those in potentially less environmentally harmful industries have a strong propensity for the rapid convergence of new voluntary reporting practices encompassing elements of social and environmental sustainability reporting.

However, according to the robustness check, while the trend in the accountability motivations of companies in potentially less environmentally harmful industries strengthens for Tokyo listed companies (i.e., manufacturing with smaller environmental impacts, information & communication, and sales & service sectors), it seems likely that accountability motivations are not exclusively confined to companies in potentially less environmentally harmful industries in the case of London listed companies (i.e., the mining, utilities, and sales & service sectors). In any case, there is a trend towards an accountability motivation in the shift to integrated reporting consistent with voluntary disclosure theory, even for London listed companies at the industry sector level.

On the other hand, Tokyo listed companies in potentially more environmentally harmful industries appear motivated to adopt integrated reporting more as a legitimizing tool. Therefore, the IIRC's proposed role for integrated reporting to be an accountability tool has not penetrated far among these companies. In other words, these companies regard integrated reporting as merely a developed form of existing selflegitimating sustainability reporting, and therefore do not have a strong incentive to differentiate their voluntary reporting philosophy using different forms of reporting. However, because Tokyo listed companies overall appear to regard integrated reporting as a transparency and accountability tool, the trend in legitimizing motivation may be limited to only a few companies. The results of the robustness check showing that only Tokyo listed companies in the mining sector have these legitimacy motivations also supports this view.

Together, these findings indicate that there are different motivations underlying voluntary reporting practices both between the two countries and between industry sectors. Overall, the main sustainabilityrelated financial transparency and accountability role of integrated reporting proposed by the IIRC has gained some traction among Tokyo listed companies but not among London listed companies. However, there is a pronounced trend towards accountability motivations in the shift to integrated reporting consistent with voluntary disclosure theory, even for London listed companies at the industry sector level.

We find we can reveal previously hidden accountability motivations by focusing on a company's voluntary reporting philosophy and practices. As a considerable amount of extant sustainability reporting research has found legitimating motivations (Cho et al., 2015), the trends demonstrated in this study appear to provide a clear distinction between the (main) role of integrated reporting (accountability) and that of sustainability reporting (legitimation), with only minor exceptions. Accordingly, voluntary disclosure theory and legitimacy theory explanations for the motivations governing corporate voluntary reporting do not conflict with each other, at least for Tokyo listed companies overall and Tokyo and London listed companies at the industry sector level, especially potentially less environmentally harmful industries. Lai et al. (2016) support this in likewise concluding that companies were not adopting integrated reporting as a legitimation strategy. This cautions against academic research regarding integrated reporting as just another form of sustainability reporting, reinforcing a point made by Humphrey et al. (2017).

#### 7. Conclusion

Through the analysis of Japanese and UK integrated reporting and

social and environmental sustainability reporting practices from the perspective of both voluntary disclosure and legitimacy theory, we provide novel empirical insights into factors motivating the shift to voluntary integrated reporting. That is, companies refine or uncover their hitherto (hidden) voluntary reporting philosophy by adapting their reporting practices, and there are trends towards accountability motivations in shifts to integrated reporting consistent with voluntary disclosure theory. Because the number of studies identifying accountability motivations, at least from the relationship between specific environmental performance and reporting, has recently been increasing, as shown in Section 3.2, our findings may be generalizable to other companies given the growth of integrated reporting.

Our findings have an important theoretical implication. The conflict between legitimacy theory and voluntary disclosure theory derives from a common belief that company stakeholder- and shareholder-oriented approaches lie in opposition. However, such conventional beliefs do not match with the current situation surrounding corporate voluntary reporting practices, thereby creating a knowledge or research gap. In this situation, as Clarkson et al. (2008) and Hummel and Schlick (2016) also suggested this possibility, our findings provide support that legitimacy theory and voluntary disclosure theory are compatible, which only becomes evident after focusing on a company's voluntary reporting philosophy and practices. Accordingly, we challenge commonly accepted beliefs and arguments in previous studies regarding the role of corporate voluntary reporting. In this sense, we provide new academic implications for future research. The point is that when analysing the role of corporate voluntary reporting, it is necessary to again appreciate that sustainability, even at the company level, consists of social, environmental, and economic sustainability.

Our findings also have a practical implication for management. The IIRC (for integrated reporting) already co-operates with other organizations that provide corporate reporting standards to ensure consistency among reporting frameworks and standards, including the GRI (for social and environmental sustainability reporting). However, despite this co-operation, a clear segregation of the roles of integrated reporting and social environmental sustainability reporting is necessary. This is because, according to our findings, companies currently actually value either the IIRC's vision for integrated reporting (especially the narrow focus on investors) or the GRI standards, depending on their reporting philosophy. In contrast, some researchers expect integrated reporting to be a future direction for sustainability reporting (Alawattage and Fernando, 2017). As previously suggested, there is a movement to replace sustainability reporting with integrated reporting. This suggests the possibility that integrated reporting could become the reporting norm over time as more companies adopt integrated reporting practices (Stubbs and Higgins, 2018). When integrated reporting becomes more ubiquitous, it may be subject to greater capture from companies seeking to use it as a new medium for legitimation. However, if the results in this study apply more broadly, then such developments in integrated reporting could challenge any legitimacy role for sustainability reporting because integrated reporting is based on the concept of "value for investors" (Flower, 2015).

In common with other quantitative studies, a limitation of this study is how well we have selected the variables as proxies for the underlying phenomena we evaluate. Although we are studying a new voluntary reporting practice, as we have used variables that are well-established proxies in studies of other areas of voluntary reporting, this is not a major limitation. However, a more in-depth study of the content and quality of integrated reports could add further insights into the motivations for individual companies initiating their integrated reporting practices.

Regrettably, it would not be possible to undertake such an analysis across the sheer number of companies covered here within the scope of a single paper. Therefore, we believe that despite these limitations, our methods are best suited to addressing this study's specific research questions. Future case study research through, for example, interviews and ethnographies, could provide more in-depth evidence of the motivations for integrated reporting or a continuation of social and environmental sustainability reporting. In addition, our methods of analysis could also apply in future empirical studies to other countries where integrated reporting is also gaining traction.

#### CRediT authorship contribution statement

**Kimitaka Nishitani:** Conceptualization, Formal analysis, Writing – original draft, Funding acquisition. **Jeffrey Unerman:** Writing – original draft. **Katsuhiko Kokubu:** Writing – review & editing, Funding acquisition.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgments

We would like to dedicate this paper to the memory of Professor Jeffrey Unerman, who passed away during the final stages of manuscript preparation. This paper forms part of the results of research supported by Grant-in-Aid for Scientific Research (B) 19H01549 from the Japan Society for the Promotion of Science, and the Environment Research and Technology Development Fund (JPMEERF16S11606) from the Environmental Restoration and Conservation Agency, Japan.

#### Appendix A. Supplementary data

Supplementary data related to this article can be found at https://doi.org/10.1016/j.jclepro.2021.129027.

#### References

- Abubakar Siddique, M., Akhtaruzzaman, M., Rashid, A., Hammami, H., 2021. Carbon disclosure, carbon performance and financial performance: international evidence. Int. Rev. Financ. Anal. 75, 101734.
- Acar, M., Temiz, H., 2020. Empirical analysis on corporate environmental performance and environmental disclosure in an emerging market context. Int. J. Emerg. Mark. 15 (6), 1061–1082.
- Ackers, B., Eccles, N.S., 2015. Mandatory corporate social responsibility assurance practices: the case of King III in South Africa. Account. Audit. Accoun. 28 (4), 515–550.
- Aerts, W., Cormier, D., 2009. Media legitimacy and corporate environmental communication. Account. Org. Soc. 34 (1), 1–27.
- Adams, C.A., 2004. The ethical, social and environmental reporting-performance portrayal gap. Account. Audit. Account. 17 (5), 731–757.
- Adams, S., Simnett, R., 2011. Integrated reporting: an opportunity for Australia's not-forprofit sector. Aust. Account. Rev. 21 (3), 292–301.
- Alawattage, C., Fernando, S., 2017. Postcoloniality in corporate social and environmental accountability. Account. Org. Soc. 60, 1–20.
- Alrazi, B., de Villiers, C., van Staden, C.J., 2015. A comprehensive literature review on, and the construction of a framework for, environmental legitimacy, accountability and proactivity. J. Clean. Prod. 102, 44–57.
- Al-Tuwaijri, S.A., Christensen, T.E., Hughes, K.E., 2004. The relations among environmental disclosure, environmental performance, and economic performance: a simultaneous equations approach. Account. Org. Soc. 29 (5–6), 447–471.
- Arunachalam, M., McLachlan, A., 2015. Accountability for business ethics in the context of financial markets authority's corporate governance principles. New. Zeal. J. Appl. Bus. Res. 13 (1), 19–34.
- Awan, U., Kraslawski, A., Huiskonen, J., 2018. Governing interfirm relationships for social sustainability: the relationship between governance mechanisms, sustainable collaboration, and cultural intelligence. Sustainability–Basel 10 (12), 4473.
- Awan, U., Kraslawski, A., Huiskonen, J., 2020. Progress from blue to the green world: multilevel governance for pollution prevention planning and sustainability. In: Hussain, C. (Ed.), Handbook of Environmental Materials Management. Springer, Cham, pp. 1–22.
- Awan, U., Sroufe, R., Kraslawski, A., 2019. Creativity enables sustainable development: supplier engagement as a boundary condition for the positive effect on green innovation. J. Clean. Prod. 226, 172–185.
- Bebbington, J., Unerman, J., O'Dwyer, B. (Eds.), 2014. Sustainability Accounting and Accountability, second ed. Routledge, Abingdon.

- Barth, M.E., Cahan, S.F., Chen, L., Venter, E.R., 2017. The economic consequences associated with integrated report quality: capital market and real effects. Account. Org. Soc. 62, 43–64.
- Berthelot, S., Cormier, D., Magnan, M., 2003. Environmental disclosure research: review and synthesis. J. Account. Lit. 22, 1–44.
- Bewley, K., Li, Y., 2000. Disclosure of environmental information by Canadian manufacturing companies: a voluntary disclosure perspective. In: Jaggi, B., Freedman, M. (Eds.), Advances in Environmental Accounting & Management. Emerald Group Publishing Limited, Bingley, pp. 201–226.
- Bourke, J.G., Izadi, J., Olya, H.G., 2020. Failure of play on asset disposals and share buybacks: application of game theory in the international hotel market. Tourism Manag. 77, 103984.
- Brundtland, G.H., 1987. Report of the world commission on environment and development: our common future. World Commission on Environment and Development. United Nations, New York.
- Cheng, L., 2017. Organized labor and debt contracting: firm-level evidence from collective bargaining. Account. Rev. 92 (3), 57–85.
- Cheng, M., Green, W., Conradie, P., Konishi, N., Romi, A., 2014. The international integrated reporting framework: key issues and future research opportunities. J. Int. Financ. Manag. Account. 25 (1), 90–119.
- Cheng, Y., Awan, U., Ahmad, S., Tan, Z., 2021. How do technological innovation and fiscal decentralization affect the environment? a story of the fourth industrial revolution and sustainable growth. Technol. Forecast. Soc. 162, 120398.
- Cho, C.H., Freedman, M., Patten, D.M., 2012. Corporate disclosure of environmental capital expenditures: a test of alternative theories. Account. Audit. Account. 25 (3), 486–507.
- Cho, C.H., Laine, M., Roberts, R.W., Rodrigue, M., 2015. Organized hypocrite, organizational façades, and sustainability reporting. Account. Org. Soc. 40, 78–94. Cho, C.H., Roberts, R.W., Patten, D.M., 2010. The language of US corporate
- environmental disclosure. Account. Org. Soc. 35 (4), 431–443. Cho, C.H., Patten, D.M., 2007. The role of environmental disclosures as tools of
- legitimacy: a research note. Account. Org. Soc. 32 (7), 639–647. Clarkson, P.M., Li, Y., Richardson, G.D., Vasvari, F.P., 2008. Revisiting the relation
- Clarkson, P.M., Li, Y., Richardson, G.D., Vasvari, F.P., 2008. Revisiting the relation between environmental performance and environmental disclosure: an empirical analysis. Account. Org. Soc. 33 (4–5), 303–327.
- Clarkson, P.M., Overell, M.B., Chapple, L., 2011. Environmental reporting and its relation to corporate environmental performance. Abacus 47 (1), 27–60.
- Cong, Y., Freedman, M., Park, J.D., 2020. Mandated greenhouse gas emissions and required SEC climate change disclosures. J. Clean. Prod. 247, 119111.
- Cormier, D., Gordon, I.M., Magnan, M., 2004. Corporate environmental disclosure: contrasting management's perceptions with reality. J. Bus. Ethics 49 (2), 143–165.
- Cormier, D., Gordon, I.M., 2001. An examination of social and environmental reporting strategies. Account. Audit. Account. 14 (5), 587–617.
- Cormier, D., Magnan, M., 1999. Corporate environmental disclosure strategies: determinants, costs and benefits. J. Account. Audit Finance 14 (4), 429–451.
- Correa-Garcia, J.A., Garcia-Benau, M.A., Garcia-Meca, E., 2020. Corporate governance and its implications for sustainability reporting quality in Latin American business groups. J. Clean. Prod. 260, 121142.
- Cox, D.R., 1972. Regression models and life tables. J. Roy. Stat. Soc. B 34 (2), 187–220. Craswell, A.T., Taylor, S.L., 1992. Discretionary disclosure of reserves by oil and gas
- companies: an economic analysis. J. Bus. Finance Account. 19 (2), 295–308. da Rocha Garcia, E.A., de Carvalho, G.M., Boaventura, J.M.G., de Souza Filho, J.M., 2021. Determinente o compareta escolar performance diseleutre a literature a reit
- 2021. Determinants of corporate social performance disclosure: a literature review.
  Soc. Responsib. J. 17 (4), 445–468.
  de Villiers, C., Rinaldi, L., Unerman, J., 2014. Integrated reporting: insights, gaps and an
- agenda for future research. Account. Audit. Account. 27 (7), 1042–1067.
- de Villiers, C., van Staden, C.J., 2011. Where firms choose to disclose voluntary environmental information. J. Account. Publ. Pol. 30 (6), 504–525.
- Deegan, C., 2002. Introduction: the legitimising effect of social and environmental disclosures: a theoretical foundation. Account. Audit. Account. 15 (3), 282–311.
- Deegan, C., 2014. An overview of legitimacy theory as applied within the social and environmental accounting literature. In: Bebbington, J., Unerman, J., O'Dwyer, B. (Eds.), Sustainability Accounting and Accountability, second ed. Routledge, Abingdon, pp. 248–272.
- Deegan, C., Rankin, M., Tobin, J., 2002. An examination of the corporate social and environmental disclosures of BHP from 1983-1997: a test of legitimacy theory. Account. Audit. Account. 15 (3), 312–343.
- Deegan, C., Unerman, J., 2011. Financial Accounting Theory, 2nd European ed. McGraw Hill, Berkshire.
- Deephouse, D.L., Carter, S.M., 2005. An examination of differences between organizational legitimacy and organizational reputation. J. Manag. Stud. 42 (2), 329–360.
- Depoers, F., 2000. A cost benefit study of voluntary disclosure: some empirical evidence from French listed companies. Eur. Account. Rev. 9 (2), 245–263.
- Dilling, P.F., Harris, P., 2018. Reporting on long-term value creation by Canadian companies: a longitudinal assessment. J. Clean. Prod. 191, 350–360.
- Dye, R.A., 1985. Disclosure of nonproprietary information. J. Account. Res. 23 (1), 123–145.
- Dye, R.A., 2001. An evaluation of "essays on disclosure" and the disclosure literature in accounting. J. Account. Econ. 32 (1), 181–235.
- Eccles, R.G., Krzus, M.P., 2010. One report integrated reporting for a sustainable strategy. John Wiley & Sons, New York.
- Eccles, R.G., Krzus, M.P., Solano, C., 2019. A comparative analysis of integrated reporting in ten countries. Accessed on. <u>https://ssrn.com/abstract=3345590/.</u> (Accessed 2 March 2020).

#### K. Nishitani et al.

Elkington, J., 2004. Enter the triple bottom line. In: Henriques, A., Richardson, J. (Eds.), The Triple Bottom Line. Routledge, London, pp. 23–38.

Epstein, M.J., Freedman, M., 1994. Social disclosure and the individual investor. Account. Audit. Accoun. 7 (4), 94–109.

- Flower, J., 2015. The international integrated reporting council: a story of failure. Crit. Perspect. Account. 27, 1–17.
- Frias-Aceituno, J.V., Rodríguez-Ariza, L., García-Sánchez, I.M., 2013. Is integrated reporting determined by a country's legal system? An exploratory study. J. Clean. Prod. 44, 45–55.
- Gianfelici, C., Casadei, A., Cembali, F., 2018. The relevance of nationality and industry for stakeholder salience: an investigation through integrated reports. J. Bus. Ethics 150 (2), 541–558.
- Giannarakis, G., Konteos, G., Sariannidis, N., Chaitidis, G., 2017a. The relation between voluntary carbon disclosure and environmental performance: the case of S&P 500. Int. J. Law Manag 59 (6), 784–803.
- Giannarakis, G., Zafeiriou, E., Sariannidis, N., 2017b. The impact of carbon performance on climate change disclosure. Bus. Strat. Environ. 26 (8), 1078–1094.
- Giannarakis, G., Zafeiriou, E., Sariannidis, N., Efthalitsidou, K., 2016. Determinants of dissemination of environmental information: an empirical survey. J. Bus. Econ. Manag. 17 (5), 749–764.
- Grassmann, M., 2021. The relationship between corporate social responsibility expenditures and firm value: the moderating role of integrated reporting. J. Clean. Prod. 285, 124840.
- Gray, R., Kouhy, R., Lavers, S., 1995. Corporate social and environmental reporting: a review of the literature and a longitudinal study of UK disclosure. Account. Audit. Account. 8 (2), 47–77.
- Greene, W.H., 2003. Econometric Analysis, fifth ed. Prentice Hall, New Jersey.
- Green, W.J., Cheng, M.M., 2019. Materiality judgments in an integrated reporting setting: the effect of strategic relevance and strategy map. Account. Org. Soc. 73, 1–14.
- Guidry, R.P., Patten, D.M., 2012. Voluntary disclosure theory and financial control variables: an assessment of recent environmental disclosure research. Account. Forum 36 (2), 81–90.
- Guthrie, J., Parker, L.D., 1989. Corporate social reporting: a rebuttal of legitimacy theory. Account. Bus. Res. 19 (76), 343–352.
- Hahn, R., Kühnen, M., 2013. Determinants of sustainability reporting: a review of results, trends, theory, and opportunities in an expanding field of research. J. Clean. Prod. 59, 5–21.
- Haider, M.B., Kokubu, K., 2015. Assurance and third-party comment in sustainability reporting in Japan: a descriptive study. Int. J. Environ. Sustain Dev. 14 (3), 207–230. Harte, G., Lewis, L., Owen, D., 1991. Ethical investment and the corporate reporting
- function. Crit. Perspect. Account. 2 (3), 227–253. Haugh, H.M., Talwar, A., 2010. How do corporations embed sustainability across the
- organization? Acad. Manag. Learn. Educ. 9 (3), 384–396. Healy, P.M., Palepu, K.G., 2001. Information asymmetry, corporate disclosure, and the
- realy, P.M., Palepu, K.G., 2001. Information asymmetry, corporate disclosure, and the capital markets: a review of the empirical disclosure literature. J. Account. Econ. 31 (1), 405–440.
- Heckman, J., 1979. Sample selection bias as a specification error. Econometrica 47 (1), 153–161.
- Hofstede, G., 2001. Culture's Consequences: Comparing Values, Behaviors, Institutions, and Organizations across Nations, second ed. Sage, Thousand Oaks.
- Hooks, J., van Staden, C.J., 2011. Evaluating environmental disclosures: the relationship between quality and extent measures. Br. Account. Rev. 43 (3), 200–213.
- Hughes, S.B., Anderson, A., Golden, S., 2001. Corporate environmental disclosures: are they useful in determining environmental performance? J. Account. Publ. Pol. 20 (3), 217–240.
- Hummel, K., Schlick, C., 2016. The relationship between sustainability performance and sustainability disclosure: reconciling voluntary disclosure theory and legitimacy theory. J. Account. Publ. Pol. 35 (5), 455–476.
- Humphrey, C., O'Dwyer, B., Unerman, J., 2017. Re-theorizing the configuration of organizational fields: the IIRC and the pursuit of 'Enlightened' corporate reporting. Account. Bus. Res. 47 (1), 30–63.
- International Integrated Reporting Committee (IIRC), 2011. Towards integrated reporting. Communicating value in the 21st century. accessed on. http://theiirc.or g/wp-content/uploads/2011/09/integrated reporting-Discussion-Paper-2011\_spre ads.pdf/. (Accessed 12 October 2015).
- International Integrated Reporting Council (IIRC), 2013. The International <integrated reporting> Framework. accessed on. http://integratedreporting.org /wp-content/uploads/2015/03/13-12-08-THE-INTERNATIONAL-integrated reporting-FRAMEWORK-2-1.pdf/. (Accessed 5 October 2015).
- International Integrated Reporting Council (IIRC), 2016. When? Advocate for Global Adoption accessed on. http://integratedreporting.org/when-advocate-for-global-adoption/. (Accessed 11 August 2017).
- Jensen, J.C., Berg, N., 2012. Determinants of traditional sustainability reporting versus integrated reporting. an institutionalist approach. Bus. Strat. Environ. 21 (5), 299–316.
- Kang, H., Gray, S.J., 2019. Country-specific risks and geographic disclosure aggregation: voluntary disclosure behaviour by British multinationals. Br. Account. Rev. 51 (3), 259–276.
- Kent, P., Ung, K., 2003. Voluntary disclosure of forward-looking earnings information in Australia. Aust. J. Manag. 28 (3), 273–285.
- Kokubu, K., Nishitani, K., Shinohara, A., Kitada, H., 2012. Environmental disclosure in Japanese companies: stakeholders influence and their information needs (in Japanese). Sangyo Keiri 71 (4), 51–61.
- Lai, A., Melloni, G., Stacchezzini, R., 2016. Corporate sustainable development: is 'integrated reporting' a legitimation strategy? Bus. Strat. Environ. 25, 165–177.

- Journal of Cleaner Production 322 (2021) 129027
- Lang, M., Lundholm, R., 1993. Cross-sectional determinants of analyst ratings of corporate disclosures. J. Account. Res. 31 (2), 2455–3271.
- Lee, S., Xiao, Q., 2011. An examination of the curvilinear relationship between capital intensity and firm performance for publicly traded US hotels and restaurants. Int. J. Contemp. Hospit. Manag. 23 (6), 862–880.
- Li, Y., Richardson, G.D., Thornton, D.B., 1997. Corporate disclosure of environmental liability information: theory and evidence. Contemp. Account. Res. 14 (3), 435–474.
- Li, D., Zhao, Y., Sun, Y., Yin, D., 2017. Corporate environmental performance, environmental information disclosure, and financial performance: evidence from China. Hum. Ecol. Risk Assess. 23 (2), 323–339.
- Liesen, A., Hoepner, A.G., Patten, D.M., Figge, F., 2015. Does stakeholder pressure influence corporate GHG emissions reporting? empirical evidence from Europe. Account. Audit. Account. 28 (7), 1047–1074.
- Lindblom, C., 1993. The Implications of Organizational Legitimacy for Corporate Social Performance and Disclosure. Paper presented at the Critical Perspectives on. Accounting Conference, New York.
- Lu, J., Wang, J., 2021. Corporate governance, law, culture, environmental performance and CSR disclosure: a global perspective. J. Int. Financ. Mark. I. 70, 101264.
- Luo, L., 2019. The influence of institutional contexts on the relationship between voluntary carbon disclosure and carbon emission performance. Account. Finance 59 (2), 1235–1264.
- Maas, K., Schaltegger, S., Crutzen, N., 2016. Integrating corporate sustainability assessment, management accounting, control, and reporting. J. Clean. Prod. 136, 237–248.
- Magness, V., 2006. Strategic posture, financial performance and environmental disclosure: an empirical test of legitimacy theory. Account. Audit. Account. 19 (4), 540–563.
- Magness, V., 2008. Who are the stakeholders now? an empirical examination of the Mitchell, Agle, and Wood theory of stakeholder salience. J. Bus. Ethics 83 (2), 177–192.
- McSweeney, B., 2002a. Hofstede's identification of national cultural differences and their consequences: a triumph of faith: a failure of analysis. Hum. Relat. 55 (1), 89–118.
- McSweeney, B., 2002b. The essentials of scholarship: a reply to Geert Hofstede. Hum. Relat. 55 (11), 1363–1372.
- Melloni, G., Caglio, A., Perego, P., 2017. Saying more with less? disclosure conciseness, completeness and balance in Integrated Reports. J. Account. Publ. Pol. 36 (3), 220–238.
- Michelon, G., Pilonato, S., Ricceri, F., 2015. CSR reporting practices and the quality of disclosure: an empirical analysis. Crit. Perspect. Account. 33, 59–78.
- Miller, S.R., Thomas, D.E., Eden, L., Hitt, M., 2008. Knee deep in the big muddy: the survival of emerging market firms in developed markets. Manag. Int. Rev. 48 (6), 645–666.
- Nishitani, K., 2009. An empirical study of the initial adoption of ISO 14001 in Japanese manufacturing firms. Ecol. Econ. 68 (3), 669–679.
- Nishitani, K., 2011. An empirical analysis of the effects on firms' economic performance of implementing environmental management systems. Environ. Resour. Econ. 48 (4), 569–586.
- Nishitani, K., Haider, M.B., Kokubu, K., 2020. Are third-party assurances preferable to third-party comments for promoting financial accountability in environmental reporting? J. Clean. Prod. 248, 119199.
- O'Donovan, G., 2002. Environmental disclosures in the annual report: extending the applicability and predictive power of legitimacy theory. Account. Audit. Account. 15 (3), 344–371.
- Pattanayak, M., 2009. Disentangling the performance and entrenchment effect of family shareholding: a study of Indian corporate governance. J. Entrepren. Financ. 12 (4), 105–128.
- Patten, D.M., 1991. Exposure, legitimacy, and social disclosure. J. Account. Publ. Pol. 10 (4), 297–308.
- Patten, D.M., 2000. Changing Superfund disclosure and its relation to the provision of other environmental information. In: Jaggi, B., Freedman, M. (Eds.), Advances in Environmental Accounting & Management. Emerald Group Publishing Limited, Bingley, pp. 101–121.
- Patten, D.M., 2002. The relation between environmental performance and
- environmental disclosure: a research note. Account. Org. Soc. 27 (8), 763–773. Perego, P., Kennedy, S., Whiteman, G., 2016. A lot of icing but little cake? Taking integrated reporting forward. J. Clean. Prod. 136, 53–64.
- Pistoni, A., Songini, L., Bavagnoli, F., 2018. Integrated reporting quality: an empirical analysis. Corp. Soc. Responsib. Environ. Manag. 25 (4), 489–507.
- Rezaee, Z., Tuo, L., 2017. Voluntary disclosure of non-financial information and its association with sustainability performance. Adv. Account. 39, 47–59.
- Rinaldi, L., Unerman, J., de Villiers, C., 2018. Evaluating the integrated reporting journey: insights, gaps and agendas for future research. Account. Audit. Account. 31 (5), 1294–1318.
- Rockness, J., Williams, P.F., 1988. A descriptive study of social accounting responsibility mutual funds. Account. Org. Soc. 13 (4), 397–411.
- Scheyvens, R., Banks, G., Hughes, E., 2016. The private sector and the SDGs: the need to move beyond 'business as usual'. Sustain. Dev. 24 (6), 371–382.
- Shocker, A.D., Sethi, S.P., 1973. An approach to incorporating societal preferences in developing corporate action strategies. Calif. Manag. Rev. 15 (4), 97–105.
- Silva–Gao, L., 2012. The disclosure of environmental capital expenditures: evidence from the electric utility sector in the USA. Corp. Soc. Responsib. Environ. Manag. 19 (4), 240–252.
- Skinner, D.J., 1994. Why firms voluntarily disclose bad news. J. Account. Res. 32 (1), 38–60.

#### K. Nishitani et al.

Stacchezzini, R., Melloni, G., Lai, A., 2016. Sustainability management and reporting: the role of integrated reporting for communicating corporate sustainability management. J. Clean. Prod. 136, 102–110.

- Stubbs, W., Higgins, C., 2014. Integrated reporting and internal mechanisms of change. Account. Audit. Account. 27 (7), 1068–1089.
- Stubbs, W., Higgins, C., 2018. Stakeholders' perspectives on the role of regulatory reform in integrated reporting. J. Bus. Ethics 147 (3), 489–508.
- Suchman, M.C., 1995. Managing legitimacy: strategic and institutional approaches. Acad. Manag. J. 20 (3), 571–610.
- Uyar, A., Karaman, A.S., Kilic, M., 2020. Is corporate social responsibility reporting a tool of signaling or greenwashing? evidence from the worldwide logistics sector. J. Clean. Prod. 253, 119997.
- van Bommel, K., 2014. Towards a legitimate compromise? an exploration of integrated reporting in The Netherlands. Account. Audit. Accoun. 27 (7), 1157–1189.

- Verrecchia, R.E., 1983. Discretionary disclosure. J. Account. Econ. 5, 179–194.
- Verrecchia, R.E., 1990. Endogenous proprietary costs through firm interdependence. J. Account. Econ. 12 (1–3), 245–250.
- Wilmshurst, T.D., Frost, G.R., 2000. Corporate environmental reporting: a test of legitimacy theory. Account. Audit. Account. 13 (1), 10–26.
- Yonekura, A., Gallhofer, S., Haslam, J., 2012. Accounting disclosure, corporate governance and the battle for markets: the case of trade negotiations between Japan and the US. Crit. Perspect. Account. 23 (4–5), 312–331.
- Yook, K.H., Song, H., Patten, D.M., Kim, I.W., 2017. The disclosure of environmental conservation costs and its relation to eco-efficiency: evidence from Japan. Sustain. Account. Manag. Pol. J. 8 (1), 20–42.
- Zhang, Y.J., Liu, J.Y., 2020. Overview of research on carbon information disclosure. Front. Eng. Manag. 7 (1), 47–62.